

ASSESSMENT OF THE ECONOMIC IMPACT OF IMPLEMENTATION OF THE PROPOSALS CONTAINED WITHIN FSA CONSULTATION DOCUMENT **CP176** WITH REFERENCE TO INCUMBENT FUND MANAGERS' ABILITY TO COMPETE IN THE UK MARKET

APRIL 2004

CONTENTS

			i age
1.	EXEC	CUTIVE SUMMARY	6
	1.1	Introduction	6
	1.2	Our approach	6
	1.3	Size of the market for softed and bundled services	7
	1.4	Segmentation of the fund management market	7
	1.5	Potential incremental cost impact of the Proposals	8
	1.6	Net impact on customers	9
	1.7	Post-implementation strategies available to fund managers	10
	1.8	Potential impact on the UK fund management market	12
	1.9	Impact on the UK economy: An indicative assessment	12
2.	INTR	ODUCTION	14
	2.1	Background and scope	14
	2.2	Proposals contained in CP176	15
	2.3	Objectives of the study	15
	2.4	Our approach	16
	2.5	Structure of the report	16
3.	SIZE	OF THE MARKET FOR SOFTED AND BUNDLED SERVICES	18
	3.1	Introduction	18
	3.2	Commissions paid by UK fund managers	19
	3.3	Non-execution services as a proportion of commissions	20
	3.4	Estimated market sizes	22
4.	MAR	KET SEGMENTATION	25
	4.1	Introduction	25
	4.2	Form of segmentation adopted	25
	4.3	Segmentation by size	26
	4.4	Segmentation by operational focus	27
	4.5	Segmentation by operating performance	27
	4.6	Hedge Funds	28
	4.7	Summary of our segmentation model	28
5.	IMPL	EMENTATION OF PROPOSAL II	30
	5.1	Introduction	30
	5.2	Method of acquiring non-execution services	30
	5.3	Method of charging for non-execution services	32

6. POTENTIAL INCREMENTAL COST IMPACT OF CP176

Page

CONTENTS

Page

	6.1	Introduction	33
	6.2	Impact of Proposal 1 on commission levels	33
	6.3	Impact of Proposal 2 on commission levels	35
	6.4	Impact of the Proposals on the value of non-execution services purchased	38
	6.5	Other costs	42
	6.6	VAT implications of the Proposals	44
	6.7	The ability of Fund Managers to recover incremental costs from customers	46
	6.8	Potential incremental cost impact on the fund management industry	50
	6.9	Summary of potential incremental cost impact	55
7.	NET I	MPACT ON CUSTOMERS	60
	7.1	Introduction	60
	7.2	Analysis of net customer impact using fund managers' views	60
	7.3	Analysis of net customer impact from our assessment of the market	61
8.	STRA	TEGIES AVAILABLE TO FUND MANAGERS	64
	8.1	Introduction	64
	8.2	Compliance	65
	8.3	Sourcing of non-execution services from other territories	67
	8.4	Partial relocation	69
	8.5	Full relocation	70
	8.6	Sale of business	72
	8.7	Market exit	73
	8.8	Conclusions on the adoption of strategies by market segment	74
9.	POTE	NTIAL IMPACT ON THE UK FUND MANAGEMENT MARKET	75
	9.1	Approach to quantifying impact on the market	75
	9.2	Summary of impact: Funds Under Management	75
10.	IMPA	CT ON THE UK ECONOMY	78
	10.1	Introduction	78
	10.2	Methodology	78
	10.3	Calculation of industry-specific GVA	78
	10.4	Estimation of GVA lost	79
	10.5	Estimation of wider economic loss using an employment multiplier	80
APPE	ENDIX 1:	SUMMARY OF METHODS USED TO ESTIMATE THE SIZE OF THE	
	MAR	KET FOR SOFTED AND BUNDLED SERVICES	82

CONTENTS

	Page
APPENDIX 2: FUND MANAGERS' VIEWS ON THE FUTURE OF THE RESEARCH SECTOR	86
APPENDIX 3: THE FUND MANAGERS QUESTIONNAIRE	90

GLOSSARY

The following abbreviations are used in this report:						
"BP"	Basis Point					
"CFD"	Contract For Difference					
"COBS"	Conduct of Business Sourcebook					
"EAD"	Earnings After Depreciation					
"FSA"	Financial Services Authority					
"FUM"	Funds Under Management					
"GDP"	Gross Domestic Product					
"GVA"	Gross Value Added					
"HMCE"	HM Customs & Excise					
"IFSL"	International Financial Services London					
"IMA"	Investment Managers Association					
"LSE"	London Stock Exchange					
"MPIS"	Market Pricing and Information Services					
"ONS"	Office of National Statistics					
"SEC"	Securities and Exchange Commission					
"SRO"	Self Regulatory Organisation					
"TER"	Total Expense Ratio					

IMPORTANT NOTICE

This report (the "Report") has been prepared by Deloitte & Touche LLP ("Deloitte") for the FSA alone in order to inform it of incremental costs that may be incurred by the UK fund management industry as a result of the adoption of proposals contained within FSA Consultation Paper CP176.

The information contained in the Report has been obtained from a number of sources which are believed to be reliable but no independent verification of any statements has been made nor has any comment or verification been made in relation to statements or expressions of opinions made in the Report. Neither the FSA nor Deloitte accept any liability in relation to any information from third party sources or any statement of opinion set out in the Report. Readers of the Report must form their own opinion as to the nature of the information contained in the Report.

Any decision to invest, conduct business, enter or exit the markets considered in the Report should be made solely on independent advice and no information in the Report should be relied upon in any way by any third party. The Report does not constitute a recommendation to use or any endorsement of any of the markets or companies referred to in the Report. The FSA is not constrained to act in accordance with the contents of the Report or the conclusions made in it and the Report constitutes neither policy of the FSA nor guidance under the Financial Services and Markets Act 2000.

Deloitte has endeavoured to ensure that data and material in the Report are accurate but does not accept liability for any omission or error contained in the Report. The information and opinions in this Report are subject to change without notice. No representation or warranty (express or implied) is made in relation to the accuracy or completeness of any information or opinions contained in the Report.

This document is for personal use and for research. All copyright and other proprietary rights in the Report remain the property of Deloitte & Touche LLP and any rights not expressly granted in these terms are reserved.

The materials in the Report do not constitute financial or other professional advice.

Neither Deloitte nor the FSA is liable for any direct, indirect, special, incidental or consequential damages arising out of the use (or the inability to use) the material in the Report, including any action or decision taken as a result of using such material. This includes but is not limited to the loss of data or loss of profit.

Deloitte is regulated by the FSA.

1.1 Introduction

Deloitte has been commissioned by the Financial Services Authority (FSA) to assess the potential impact on the UK fund management industry of the FSA's policy proposals ("the Proposals") concerning soft commission arrangements and bundled brokerage services contained in FSA Consultation Paper CP176. The stated aim of CP176 is to increase transparency and accountability to the customers of fund managers by:

- limiting the goods and services, beyond trade execution, that can be bought using
 "soft credits"; and
- mandating that the cost of acquiring non-execution services in a package along with trade execution should not be passed through automatically by fund managers to their customers' funds.

The FSA requested that we address two possible impacts of its Proposals:

- incremental costs that may be incurred by the UK-regulated fund management industry as a result of the implementation of one or both proposals contained in CP176; and
- the extent to which incremental costs that are incurred may lead fund managers to exit the UK market, either by relocating operations serving UK customers overseas, selling assets managed in the UK, or closing their UK operations.

We have also undertaken an indicative analysis of the effect that relocation and exit decisions may have on UK GDP.

1.2 Our approach

In order to develop estimates of the source and magnitude of incremental costs that may be incurred by UK fund managers, and their potential strategies in response, we:

- were granted access to a subset of CP176 consultation responses;
- developed hypotheses relating to types of incremental costs that may be incurred by fund managers;
- developed a market segmentation model;
- developed and forwarded questionnaires to fund managers, brokers and industry associations and used these as the basis for interviews with respondents and others; and

obtained industry data from a variety of external sources.

Further detail on our approach can be found at Section 2.4 below. Information sourced in line with this approach consists of data obtained from a number of sources. While we believe this to be reliable, we have in most cases not sought to verify it. Our analysis also depends on opinions expressed to us by fund managers, brokers and others which we have in some cases sought to challenge, validate or supplement using our knowledge of the fund management market.

1.3 Size of the market for softed and bundled services

Using three alternative potential methods, we estimate the size of the market for softed goods and services in 2002/03 to be between £105m and £125m. Within this, we estimate the size of the market for softed Market Pricing and Information Services (MPIS), the focus of Proposal 1 of CP176, to be between £50m and £59m. The size of the market for bundled goods and services is estimated at between £653m and £780m.

Taking into account fund managers' predictions relating to the lower quantity and value of services purchased if the Proposals are implemented, we estimate aggregate cost savings for consumers resulting from altered purchasing patterns following implementation ranging from £61m to £116m.

See Section 3 for further detail on the above estimates.

1.4 Segmentation of the fund management market

To inform our analysis of costs impacting different types of fund management companies, we divided the market into size segments comprising:

- 1. Large fund managers: over £50bn of Funds Under Management (FUM)
- 2. Medium-sized fund managers: between £5bn and £50bn of FUM
- 3. Small fund managers: less than £5bn of FUM

For some analyses, these are further divided into the following sub-segments:

- 1. Large, UK focused
- 2. Large, non-UK focused
- 3. Medium, UK focused
- 4. Medium, non-UK focused
- 5. Small, UK focused with above average operating performance*
- 6. Small, non-UK focused with above average operating performance*
- 7. Small, UK focused with below average operating performance*
- 8. Small, non-UK focused with below average operating performance*

* Performance is measured in terms of 2002/03 margins. Derivation of margins data is discussed at Section 6.8.

When assessing strategies available to fund managers if the Proposals are implemented, we include hedge funds as an additional sub-segment in recognition that their operating models, level of charges and other factors distinguish them from other small and medium-sized fund managers.

See Section 4 for further detail on this model of segmentation.

1.5 Potential incremental cost impact of the Proposals

Using 2002/03 financial information from an FSA database containing 443 fund management companies, around 73% of the UK fund management market, summary estimates of the margins impact¹ of potential incremental costs incurred by each of the above market segments following implementation of CP176 are shown in the table below. These are based on market analysis, questionnaire returns and interviews with fund managers, brokers and industry associations regarding potential changes to commission levels and patterns of demand for non-execution services, and fund managers' ability to recover incremental costs from customers.

Table 1.1 - Summary of Deloitte analysis of potential margins impact:									
Average impact on		Margin 2002/03		Adjusted margin 2002/03		Difference (percentage points)			
		LOW	HIGH	LOW	HIGH	Ľow	, ні <i>G</i> Н		
Total market		13.47%	13.85%	8.01%	10.47%	(5.45%)	(3.38%)		
		12.30% 5.03%	12.31% 6.06%	6.71% (3.14%)	9.47% 0.30%	(5.60%) (8.17%)	(2.84%) (5.76%)		
Large non-UK		16.67%	17.03%	13.72%	15.18%	(2.95%)	(1.85%)		
Medium		15.17% 21.17%	17.32% 22.77%	11.64% 17.63%	14.46% 19.97%	(3.53%) (3.54%)	(2.86%)		
Medium non-Uk	<	12.60%	15.04%	9.08%	12.16%	(3.52%)	(2.88%)		
Small Small UK1 Small UK2 Small non-UK1 Small non-UK2		10.14% 43.22% (7.00%) 42.80% (2.14%)	13.90% 46.08% (6.94%) 45.84% (0.98%)	2.04% 32.93% (19.45%) 27.04% (7.91%)	9.47% 40.85% (12.69%) 40.09% (4.29%)	(8.10%) (10.30%) (12.45%) (15.76%) (5.77%)	(4.43%) (5.22%) (5.75%) (5.75%) (3.32%)		
Key: Small	1 = Above aver	age margins	2002/03						
Small	2 = Below avera	age margins	2002/03						
	Source: Deloitte analysis								

Margins for the industry as a whole are estimated to fall by between 3.4 and 5.2 percentage points. The least impact is anticipated on the medium segment, and the most on the small segment, where the potential impact ranges from 4.8 to 7.5 percentage points.

¹ See Section 6.8 for a detailed description of the derivation of margins data. As an element of depreciation is contained in the cost data, revenues less costs relates to what may be termed Earnings After Depreciation ("EAD") rather than operating profit, and thus margins will understate operating margins. As 2002/03 represented the nadir of the global bear market, margins analyses shown are likely to understate average industry margins over an investment cycle.

We converted the estimates above into an analysis of the number of companies that may have experienced positive and negative margins if CP176 had been implemented in 2002/03. This is shown in the table below.

Table 1.2 - Analysis of positive and negative margins based on Deloitte cost impact assessment															
	No. of firms	% of total FUM	Firms with positive margin 02/03	% of segmen t FUM	Firms with negative margin 02/03	% of segment FUM	Firn margii becon becaus	ns wh n may ne neg se of	ose / have gative CP176	% of s	egme	nt FUM	% of fin have e negat fo implen CP17 segn	rms th xperi ive m llowir nenta '6 (as nent l	hat may ienced pargin ng tion of % of FUM)
Total	443	100%	327	79%	116	21%	Worst 107	-	Best 83	Worst 16%	-	Best 5%	Worst 37%	-	Best 26%
By segment															
Large	11	46%	9	79%	2	21%	2	-	0	21%	-	0%	42%	-	21%
Large UK	6	29%	4	67%	2	33%	2	-	0	27%	-	0%	60%	-	33%
Large Non- UK	5	17%	5	100%	0	0%	0	-	0	0%	-	0%	0%	-	0%
Medium	48	42%	35	81%	13	19%	5	-	3	14%	-	5%	33%	-	24%
Medium UK	14	13%	12	94%	2	6%	1	-	1	9%	-	9%	15%	-	15%
Medium Non UK	34	29%	23	76%	11	24%	4	-	2	12%	-	2%	36%	-	27%
Small	384	13%	283	72%	101	28%	100	-	80	42%	-	32%	70%	-	60%
SmallUK1	103 104	3%	103	100%	0	0% 37%	5	-	3	9% 49%	-	7% 39%	9% 85%	-	7% 76%
SmallNon-	66	2%	66	100%	42	0%	11	-	7	23%	_	11%	23%	-	11%
UK1 SmallNon- uk2	111	4%	52	44%	59	56%	39	-	32	37%	-	29%	94%	-	86%
Key:	Small	1 = Abo	ve averaç	ge margins	s 2002/03	•									
	Small	2 = Belo	ow averag	e margins	\$ 2002/03										
				-				9	Source	: FSA	Data	abase /	Deloitt	e an	alvsis
															, 510

According to this analysis, of the 443 companies included in the FSA's database, a range of 83 to 107 out of 327 companies with a positive margin in 2002/03 may have been vulnerable to reporting a negative margin as a result of implementation of the Proposals at that snapshot in time (equal to 5% - 16% of total FUM), taking the number of firms with a negative margin up to a range of 199 to 223 (equal to 26% - 37% of total FUM).

See Section 6 for further detail on incremental costs that may be incurred by fund managers if the Proposals are implemented.

1.6 Net impact on customers

We examined the potential net impact on funds belonging to customers of fund managers if the Proposals are implemented resulting from the following factors:

- decreases to commission levels;
- changes to the size of the market for softed and bundled services; and

fund managers' ability to recover incremental costs from customers through increased management fees.

The analysis is indicative of the financial impact on customers that may result from these factors. It is not intended to account for other types of impacts that may affect some but not all customers including:

- potential increased tax liability for some investment vehicles where management fees increase;
- changes to fund managers' trading behaviour or strategy; and
- changes in market structure (e.g. consolidation).

We accepted fund managers' views relating to factors analysed resulting from Implementation of Proposal 1. This resulted in a small aggregate estimated net cost spread across the customers of all UK-regulated fund managers of between £2.0m and £2.4m a year. This arises because fund managers may be able to recover a slightly higher level of costs from customers than the level of counteracting falls in commission obtained from brokers by fund managers on their customers' behalf. However, this trade-off between commissions and cost recovery could easily swing in the opposite direction, and may also be affected by decisions to implement the Proposals simultaneously or individually, as fund managers may be more able to recover the entire value of non-execution services foregone through lower commissions (if both proposals are implemented), than to recover the value of softed services alone (if only Proposal 1 is enacted).

We estimate a net customer impact ranging from a £24.1m net loss to a £290.9m net saving per annum resulting from Implementation of Proposal 2. This is derived from:

- a smaller estimated value of currently bundled non-execution services purchased and passed through to customers following implementation of the Proposals; and
- a higher estimated fall in commissions attributable to bundled services than the anticipated level of cost recovery of these services from customers.

The estimated impact of both proposals taken together ranges from a net cost of £26.1m to a net saving of £288.4m.

See Section 7 for further detail on the potential customer impact resulting from implementation of the Proposals.

1.7 Post-implementation strategies available to fund managers

The diagram below shows strategies available to fund managers if the Proposals are implemented:



Figure 1.1 - Strategies available to Fund Managers

Strategy 1 above would be followed by fund managers in a position to fully comply with the Proposals, while strategies 2a and 2b imply a level of evasion, but should not result in assets leaving the UK market. However, strategies 2c (relocation), 3a (sale) and 3b (closure / exit) each result in the exit of FUM from the market.

In the table below we have input ranged probabilities of the likelihood of each segment and sub-segment adopting each strategy. These have emerged from discussions with fund managers, brokers and other industry stakeholders, questionnaire responses, market analysis and consultation responses sent by fund managers to the FSA. They are also influenced by the potential cost impact on each segment as detailed in Section 1.5 above.

Table 1.3 - Deloitte assessment of the probability of strategy adoption by segment								
Segment /Strategy	1. Comply	2a. Source from overseas	2b. Partially relocate	2c. Relocate	3a. Sell	3b. Exit/Close		
Adoption probabilities								
LargeUK	70% - 85%	15% - 25%	0% - 0%	0% - 0%	0% - 0%	0% - 0%		
Medium LIK	40% - 55%	40% - 50%	0% - 10%	0% - 0%	0% - 0%	0% - 0%		
MediumNon-	40% - 60%	10% - 30%	0% - 20%	5% - 15%	0% - 0%	0% - 0%		
SmallUK1	60% - 80%	0% - 10%	0% - 0%	10% - 15%	0% - 6%	0% - 3%		
SmallNon-UK1	40% - 50%	35% - 45%	0% - 0%	10% - 15%	0% - 5%	0% - 5%		
SmallUK2	40% - 70%	5% - 10%	0% - 0%	5% - 10%	5% - 20%	5% - 15%		
SmallNon-UK2	30% - 40%	15% - 25%	0% - 0%	20% - 30%	5% - 10%	5% - 20%		
Hedge funds	65% - 90%	0% - 0%	0% - 0%	5% - 25%	0% - 5%	0% - 5%		
Key: Small	Key: Small 1 = Above average margins 2002/03							
Small	2 = Below avera	age margins 20	02/03					
Source: Deloitte analysis								

See Section 8 for further detail on the strategies that may be followed by fund managers in response to implementation of the Proposals.

1.8 Potential impact on the UK fund management market

The probabilities shown above of the proportion of each market segment adopting the six possible strategies produce a range of potential FUM exit from £53bn to £142bn (2% to 5.5% of the market, estimated at £2,600bn at the end of 2002^2). This incorporates a central case estimate of £97bn of funds (3.7%) exiting.

Of this range of estimated exit, £24bn-£80bn is managed by the segment of medium-sized fund managers and £28bn-£62bn by the segment of small fund managers. The highest level of potential exit is among the small non-UK below average performance sub-segment (SmallNon-UK2), in which an estimated range of between 16.5% and 31.4% of FUM may exit.

See Section 9 for further detail on the impact on the UK fund management market resulting from the level of FUM exit estimated above.

1.9 Impact on the UK economy: An indicative assessment

A high-level indicative analysis was carried out of the potential impact on the UK economy resulting from the above estimated range of FUM exiting the UK market following implementation of the Proposals

The level of estimated exit would result in a direct loss of Gross Value Added (GVA) to the economy. An indirect loss also occurs resulting from the knock-on effects on other parts of the supply chain and related industries of a reduced level of employment in the fund management market. Taking each of these effects into account, the total level of GVA loss may be estimated at between £321m and £819m, representing between 0.035% and 0.090% of UK economy GVA (£910bn in 2002/03).

GDP is obtained from GVA by adding taxes paid to and subtracting subsidies received from the Government. As industry-level economic activity is not normally expressed in terms of GDP, the ratio of industry GVA lost to whole-economy GVA is more meaningful than comparing industry GVA with whole-economy GDP. However, for completeness, both ratios are noted in the table below which indicatively shows the impact on the UK economy of the potential level of FUM exit estimated above.

² Fund Management, IFSL, 2003.

Table 1.4 – FUM Exit as proportion of 2002/2003 GDP						
	Low case (£m)	Central case (£m)	High case (£m)			
GVA exiting the economy	321	578	819			
2002/03 GVA		909,827				
Exit as proportion of 2002/03 GVA	0.035%	0.064%	0.090			
2002/03 GDP		1,054,061				
Exit as proportion of 2002/03 GDP	0.030%	0.055%	0.078%			
Source: Deloitte analysis						

See Section 10 for further detail on our indicative assessment of the impact on the UK economy resulting from the level of FUM exit estimated above.

2.1 Background and scope

Deloitte has been commissioned by the Financial Services Authority (FSA) to assess the potential impact on the UK fund management industry of the FSA's policy proposals concerning soft commission arrangements and bundled brokerage services contained in FSA Consultation Paper CP176. The FSA has specifically requested that we address two possible impacts of its proposals:

- incremental costs that may be incurred by the UK-regulated fund management industry as a result of the implementation of one or both proposals contained in CP176; and
- the extent to which incremental costs that are incurred may lead fund management companies to exit the UK market, either by relocating operations serving UK customers overseas, selling assets managed in the UK, or closing their UK operations.

We have also undertaken an indicative analysis of the effect that relocation and exit decisions may have on UK GDP.

The scope of this report excludes:

- an analysis of costs that may fall on closely related industries (e.g. brokerage);
- the extent to which best execution and the rules governing soft commission are currently dealt with by rules contained within the FSA's Conduct of Business Sourcebook (COBS); and
- consideration of the issue of ease of implementation and enforcement of the proposals contained within CP176.

In carrying out the analysis, and following discussions with the FSA, we have conducted the study on the assumption that, if CP176 is implemented, it will apply to all fund managers regulated in the UK. Thus, the location of customers will not influence the applicability of the regulation to UK fund managers (though the locational mix of clients is likely to influence strategies followed by fund managers following implementation). Under this assumption, redomiciling funds overseas would not be an effective evasion strategy, and has therefore not been examined in detail in this report.

2.2 Proposals contained in CP176

The FSA's stated aim In the above consultation paper is to increase the transparency of transactions carried out on customers' behalf, and the accountability of fund managers to their customers by implementing the following proposals ("the Proposals"):

Proposal 1:

Limit the goods and services, beyond trade execution, that can be bought in the first instance with commission or order flow. The purchase with soft credits of Market Pricing and Information Services (MPIS), such as dealing screens, would be specifically prohibited.

As part of its consultation process, the FSA also sought views on whether the purchase of other services, such as computer hardware and software, other equipment and custody services should be prohibited using soft credits.

Proposal 2:

Mandate that the cost of acquiring non-execution services in a package along with trade execution should not be passed through automatically by fund managers to their customers' funds. This would apply in particular to the use of commission to buy investment research.

The unbundling of execution and non-execution services is not mandated under this proposal though this is one possible outcome. Alternatively, fund managers may continue to buy bundled services, rebate the non-execution portion of these to their customers' funds, and attempt to recover the cost of the purchase through a higher management fee or explicit charges.

2.3 Objectives of the study

Among themes raised by fund managers that contributed to the FSA's consultation process following its publication of CP176, it was claimed that the proposals would result in UK fund managers incurring additional costs such that the industry's international competitiveness would decline. Specifically, it was alleged that the prohibition of automatic cost pass-through to customers for softed and bundled goods and services may have such a large impact on firms' operating costs that it could make UK fund management more expensive, or otherwise less attractive, than that of other jurisdictions, particularly for overseas customers, small companies, and new entrants. As a result fund managers may be incentivised to re-locate to jurisdictions other than the UK.

The purpose of this report is to assess the potential cost impact on UK fund managers of the Proposals and whether this might be significant enough to influence decisions around the location of fund management activities in the UK.

2.4 Our approach

In order to develop estimates of the source and magnitude of incremental costs that may be incurred by UK fund managers, and their potential strategies in response, we adopted the following approach:

- we were granted access to around 50 of 150 consultation responses to CP176 that were sent to the FSA by fund managers and brokers. These were chosen by the FSA as being the most relevant and important responses, and as representing a cross-section of opinion forwarded by fund managers, brokers and others;
- in light of views on the source and magnitude of incremental costs put forward in these responses, we developed a number of cost hypotheses, each of which dealt with one or a number of potential incremental costs that might arise from implementation of the FSA's proposals;
- a market segmentation model based on size and geographic focus was developed to assess the differing potential impacts on different segments of the industry, and the strategies open to companies in each segment in response;
 - the cost hypotheses were used as the basis for the development of separate questionnaires sent to a cross-section of fund managers to ensure segmental coverage (the "fund managers questionnaire"), and to several brokers and industry associations;
 - when questionnaires were complete, each respondent was interviewed to discuss and test the information provided. Interviews were also carried out with companies recommended by the Investment Management Association (IMA) or which had contacted the FSA directly. In total, 24 fund managers were interviewed and 18 questionnaire returns were received, representing around 30% of Funds Under Management (FUM) in the UK fund management industry. Twelve other interviews were also carried out with brokers, industry associations and research companies, and;
 - industry data was obtained from a variety of external sources.

In adopting the approach set out above, the information contained in this Report relies on data obtained from a number of sources. While we believe this to be reliable, we have in most cases not sought to verify it. Our analysis also depends on opinions expressed to us by fund managers, brokers and others which we have in some cases sought to challenge, validate or supplement using our knowledge of the fund management market.

2.5 Structure of the report

The remainder of this report is structured as follows:

Section 3 contains an indicative estimate of the size of the market for softed and bundled services in the UK.

Section 4 introduces our method of segmenting the UK fund management market in order to assess the ways in which incremental costs may affect each segment, and the strategies available to each segment in response to the Proposals.

Section 5 examines the potential mechanism of implementation of Proposal 2 of CP176, in particular the way that non-execution services may be purchased by fund managers and charged to customers.

Section 6 considers the potential incremental cost impact of the Proposals on each segment of the industry.

Section 7 assesses the potential net impact of the Proposals on customers' funds.

Section 8 examines the post-implementation strategies available to each segment of the industry, and the probability of adoption of each strategy on a segmental basis.

Section 9 assesses the potential impact on the fund management industry in terms of funds exiting the UK from adoption of strategies presented in Section 8.

Section 10 extends the analysis presented in Section 9 to arrive at an indicative estimate of the potential impact on the UK economy.

3.1 Introduction

In this section, we estimate the size of the UK market for MPIS as a subset of softed services, total softed services and bundled services. This analysis has been undertaken for three reasons:

- to inform the FSA as it seeks to fulfil its duty to take account of the principle of proportionality before implementing the Proposals;
- to inform and provide a check on our analysis of the potential cost impact resulting from implementation of the Proposals; and
- to provide a framework to facilitate the updating from 2000 to 2003 of the cost benefit analysis relating to the Proposals undertaken by OXERA.³

The ranged estimate of market sizes given below should be seen as indicative due to a lack of available data that can be used to undertake a full analysis. We have estimated market sizes by using a "top-down" approach which:

- utilises three alternative methods to estimate the level of worldwide commissions paid by UK fund managers to brokers in 2002/03; and
- applies to this estimate the level of MPIS, total softed services and bundled services purchased as a proportion of total commissions.

We were unable to supplement this approach with a "bottom-up" analysis of the market. This might have been achieved by obtaining the value for MPIS provided to fund managers under softing arrangements by the main providers of these services. The total value of softed and bundled services could then be imputed from the value of MPIS by assessing the relative value of these services as a proportion of total commissions. This approach was not taken forward as:

- no definition of MPIS is currently available⁴;
- feedback was received from providers of information services that the term MPIS may largely relate to low-value added data feeds previously provided to fund managers, rather than some of the smart research tools now on the market; and

^{3 &}quot;Cost-benefit analysis of the FSA's policy proposals on soft commission and bundling", OXERA (2003).

⁴ We have not sought to ascribe a definition to MPIS for the purposes of market sizing and assessing cost impacts of the Proposals in this report. We have instead relied on fund managers' perceptions of the value of MPIS that they receive as a proportion of the total value of services provided to them.

the fragmentation of the market caused by increased provision of data services by investment banks⁵ made the estimation of a total market value unrealistic in the time available.

Section 3.2 gives a summary of methods used to estimate total commissions paid by fund managers. Appendix 1 gives further detail on these approaches. Section 3.3 estimates the level of commissions attributable to MPIS, softed and bundled services as a proportion of total commissions. Section 3.4 combines these analyses to arrive at a ranged estimate of market sizes. It also presents an estimate of aggregate cost savings that may accrue from a fall in demand for these services if the Proposals are implemented as drafted.

3.2 Commissions paid by UK fund managers

We have used three methods to arrive at an estimate of worldwide commissions paid by UK fund managers in 2002/03. These are described in summary below and in detail at Appendix 1:

- Method 1: Value of equity trades the total value of equity trades carried out for UK fund managers was estimated using data on the value of trades in UK equities from the London Stock Exchange (LSE), and data relating to the composition of managed equity funds per world trading region (e.g. Europe / US) from the IMA. This was then multiplied by average UK and overseas commissions data from Elkins McSherry to arrive at an estimate of UK fund managers' total worldwide commission payments.
- Method 2: Stamp duty take statistics on 2002/03 stamp duty receipts from equity transactions were obtained from the Inland Revenue and used to impute 2003 commissions from UK-traded equities. Commissions relating to overseas equity trades were estimated as in the method above and added to UK commissions to arrive at a worldwide commissions total.
- Method 3: Fund activity analysis a WM Company survey of the level of trading activity of pension fund equity holdings by world trading region was applied to IFSL and IMA statistics relating to total equity funds under management in the UK to arrive at a total value of worldwide trades placed for UK fund managers. As with the methods above, this was then multiplied by average commission levels to obtain a worldwide commissions total.

⁵ For example, the acquisition in April 2004 of Barra by Morgan Stanley Capital International Inc.

3.3 Non-execution services as a proportion of commissions

In order to estimate the size of the market for MPIS, softed and bundled services, the proportion of total commissions attributable to these services has been estimated. This allocation has been taken from the fund managers questionnaire which asked fund managers to assess the level of total commissions paid to brokers spent on:

- ' execution;
- working a trade;
- softed services; and
- bundled services.

Working a trade may be loosely described as the timed placement of different parts of a single order to achieve best execution. As such, it is generally considered to be inseparable from trade execution rather than a service bundled with it. To the extent that some brokers offer commission rates for execution-only services and a separate rate including the active working of trades, working a trade may be considered one of a menu of options offered alongside trade execution and thus on the broadest definition, part of a wider bundle. However, in the remainder of this report we have not included working a trade within our definition of the bundle as this aspect of the trade cannot be provided by a party other than that which executes the trade and so cannot be unbundled in the same way as other (non-execution) services provided by the broker.

Questionnaire respondents' division of commissions paid to brokers into the categories above are shown in Table 3.1. This splits out commission paid to UK and overseas brokers⁶, shows an average weighted by the value of commissions paid to each, and compares the results with an earlier analysis undertaken by OXERA.⁷

Table 3.1 Components of commission spend								
bps	OXERA	UK brokers	Overseas brokers	Weighted average				
Average commission Execution only Working a trade Soft commission Bundled services	14.0 5-8 2-5 1.0 3.0	14.9 7.0 4.7).5 2.7	21.2 7.7 5.6 1.3 3.6	16.6 7.2 4.9 0.7 3.8				
Source: Deloitte fund managers questionnaire								

⁶ Overseas brokers will often be used to execute trades in overseas equities. Fund managers were asked for this split to gain an idea of the extent to which commission levels tended to differ between trades that take place in the UK and those executed overseas. Overseas trades may be executed in less liquid markets than the UK, raising commission levels. However, respondents' estimates indicate that they believe a higher value of bundled services also results from these trades.

⁷ "An assessment of soft commission arrangements and bundled brokerage services in the UK", OXERA (2003).

The table shows a good deal of similarity between our findings and those of OXERA, with the exception of the higher weighted average commission rate paid. The 0.8bp attributable to soft credits can be further split out between the goods and services below, as extracted from fund managers' responses to the questionnaire.



Respondents also estimated the proportion of bundled services attributable to research and other services.



The above levels of MPIS, softed and bundled services received are summarised as a percentage of total commissions in Table 3.2 below.

Source: Deloitte fund managers questionnaire					
Total	100%				
Execution/working a trade	73%				
Total non-execution	27%				
Bundled services	23%				
Other soft commission	2%				
MPIS	2%				
Proportion attributable to:					
	Percentage of total				
non-execution services					
Table 3.2 Proportion of commis	Table 3.2 Proportion of commission spend on				

The value of MPIS and softed services can be established from the value of commissions attributable to these services by dividing by an average multiple which reflects the difference between services received and commissions paid. We have applied a multiple of 1.17 to arrive at the value of MPIS and softed services received.⁸

3.4 Estimated market sizes

Table 3.3 below shows estimates of the size of the markets for MPIS, softed and bundled services in 2002/03 by applying the proportion of commissions attributable to these services estimated at Section 3.3 to total worldwide commissions paid by fund managers estimated at Section 3.2.

Table 3.3 Estimated size of the markets for MPIS, total softed and bundled services in 2002/03							
	T ()	Value	£m				
	commissions (UK and overseas)	MPIS	Softed services	Bundled services			
Proportion of commission attributable to Softing multiplier		2% 1.17	2% 1.17	23%			
Commissions estimation method:							
1. Value of equity trades	2.878	50	105	653			
2. Stamp duty take	3.334	58	122	757			
3. Fund activity analysis	3,435	59	125	780			
Sources: Fund managers guestionnaire/LSE/IMA/IFSL/Inland Revenue/Elkins McSherry/WM Company							

The low-case and high-case estimates above can be used to form a range of market sizes of $\pounds 50m - \pounds 59m$ (MPIS), $\pounds 105m - \pounds 125m$ (softed services) and $\pounds 653m - \pounds 780m$ (bundled services). This produces an estimated size of the market for softed and bundled services ranging from $\pounds 758m$ to $\pounds 905m$, equivalent to 26.3% of estimated total commissions which range from $\pounds 2,878m$ to $\pounds 3,435m$.

⁸ This is taken from the OXERA report "An assessment of soft commission arrangements and bundled brokerage services in the UK", OXERA, (2003). This multiple applies to 2001. OXERA found the 2000 multiple to be 1.15, implying that it may be relatively stable over time.

The ranges of the size of the markets for MPIS and total softed services are below those estimated by OXERA in its cost-benefit analysis of the Proposals using 2000 commissions data. OXERA estimated a market for softed services of £160m of which £90m was attributable to MPIS. The ranged value of bundled services overlaps OXERA's estimated range of £500m - £720m.

Decreases in the volume and value of trading between 2000 and 2003 are likely to have had a significant downward impact on commissions paid and thus the market sizes for the above services since 2000. Our analysis partly offsets this effect by attempting to take account of trades in overseas equities placed away from the UK, which were not included in OXERA's estimates.

3.4.1 Aggregate cost savings resulting from changes in demand and price

Any fall in demand for non-execution services, and thus the value purchased, if Proposals 1 and 2 were implemented as drafted (e.g. Proposal 1 to incorporate a ban on the softing of MPIS only) results in aggregate cost savings to the purchasers of those services. As the Proposals aim to alter the automatic pass-through of the price of bundled services to consumers, this cost saving should accrue to consumers but will be dependent on the extent to which the value of services purchased can be recovered from them by fund managers. Less then 100% recovery will result in increased costs for fund managers and cost savings for consumers while more than 100% recovery will decrease fund managers' costs and increase consumer costs, offsetting any fall in the value of services purchased.

Questionnaire respondents forecast that, if the Proposals are implemented, an average 4% fall in the quantity of softed services purchased would result⁹, while changes in price should not occur as softed services are generally purchased direct from suppliers in the UK, rather than bought in bulk by brokers, as in the United States. This reduction in demand results in an estimated range of cost savings to fund managers of between £2.0m and £2.4m (MPIS), and a range of £4.2m to £5.0m (all softed services).

Respondents were also asked to indicate the anticipated change in price and quantity of bundled services received following implementation of the Proposals.¹⁰ Of those that were willing to estimate both, a 15% drop in the value of services received was reported on average. Of those that indicated changes in price or quantity but gave no answer for the other variable, and if no change is assumed for this variable, a 9% drop in the value of services received results. If these changes in demand and price are used to re-size the market for bundled services, cost savings of between £59m and £114m result. Total cost

^{9 20%} of respondents predicted an average 20% fall in quantity of services purchased. 20% x 20% = 4%.

¹⁰ Price and quantity effects could occur in the same direction or different directions. Results are also complicated by 2nd order affects resulting from strategies adopted by fund managers following changed conditions in the research market (e.g. greater use of in-house research to gain a competitive advantage).

savings resulting from falls in demand forecast by fund managers are estimated at between $\pounds 61m$ and $\pounds 116m$.

4. MARKET SEGMENTATION

4.1 Introduction

We have separated the UK-regulated fund management market into segments in recognition that it is not homogenous, but it is also not feasible to consider every fund manager in the UK individually. However, by grouping fund managers with similar characteristics into segments and sub-segments, each of these can be analysed separately to judge:

- potential costs arising from the Proposals, which will have a differential impact on different types of company (e.g. those with the smallest cost bases at present may have the greatest difficulties in absorbing extra fixed costs);
- the range of strategies, from compliance to market exit, available to each segment; and
- strategy selection from this range, which will differ by type of company and their position within the market.

4.2 Form of segmentation adopted

We segmented the market by the following characteristics:

- ' size;
- operational focus; and
- (for the small segment) operating performance.

In addition, hedge funds were analysed separately from other funds to take account of the different model of operations and charges employed by, and different strategies available to, these firms.

Segmentation by type of customer was also considered. For example, retail customers might be expected to respond to the Proposals in a different way to institutional customers; and life company customers may respond differently to pension fund customers. However, this was rejected as:

- around 94% of the customers of UK-regulated fund managers are institutional¹¹; and
- a high degree of correlation was found to exist between type of customer and size
 (e.g. the largest fund managers tend to manage the bulk of the assets of the largest
 institutional customers).

MARKET SEGMENTATION

4

The definition of the segments was informed by consultation responses to CP176, interviews with fund managers and responses to the fund managers questionnaire. The allocation of individual companies to segments was generated using the FSA database of fund managers ("the FSA database"), and topped up with other publicly available data.¹²

4.3 Segmentation by size

Our primary form of market segmentation is by size and this was the main means of selection of the companies to which the fund managers questionnaire was sent. The other forms of segmentation described in this section are secondary to segmentation by size and thus compose sub-segments of the market.

If the Proposals are implemented, there are a number of reasons to believe that larger fund managers would be affected (and would act) differently to their smaller counterparts:

- it is believed by many that softing and bundling in the UK market results in the subsidisation of smaller fund mangers by their larger peers. Softing and bundling are also said to support the establishment of new fund managers by reducing the costs of start-up, or passing a proportion of these through to customers. Implementation of the Proposals may reduce or end such cross-subsidies, exacerbating any negative cost impact on smaller fund managers and new entrants;
- additional costs to fund managers arising from introduction of the Proposals may be less significant (as a proportion of total costs or FUM) for large fund managers compared to their smaller peers, and could therefore be more easily absorbed and/or represent a smaller proportionate impact on customers' fees;
- the size of a fund manager has some impact on its ability to re-negotiate competitive commissions with brokers and fees with customers; and
- smaller fund managers should be more flexible making it easier for them to quickly put into effect policy decisions which affect the whole business.

For all of these reasons, size may be the most significant factor determining the impact of the Proposals on individual fund management companies. We have allocated fund managers to three segments on the basis of FUM in 2002/03:

- large: greater than £50bn;
- medium: between £5bn and £50bn; and
 - small: less than £5bn.

¹¹ Fund Management Survey, IMA (2002).

¹² Most notably, Money Management, July 2003

4 **MARKET SEGMENTATION**

The boundaries for these segmentation criteria were informed by interviews with fund managers and analysis of questionnaire responses.

4.4 Segmentation by operational focus

Relocation of operations to a territory outside the UK may be a feasible potential strategy available to UK-regulated companies if the Proposals are implemented as drafted. The ease with which this can be achieved will depend on a number of criteria, including:

- location of customer base fund managers with a predominance of overseas customers may find it easier to switch their operations to another territory, and may risk losing some overseas customers if they do not do so. However, a fund manager with mainly UK customers may find it harder to explain to them why it is proposing to move away from the domestic market;
- Iocation of offices a global player may have offices in several continents and may therefore find it relatively easy to switch operations overseas, or to find other strategies to avoid or reduce the impact of CP176 on their costs and/or operations; and
- location of principal ownership a fund manager with an overseas-domiciled parent will have access to overseas resources, and its parent may consider the UK a relatively minor part of its wider operations.

We have used the above criteria to inform our judgement of whether companies' operational focus is UK or overseas based. This allocation was informed by publicly available data, discussions with fund managers and our knowledge of the market.

4.5 Segmentation by operating performance

The Proposals may have the greatest impact, in terms of pressure on margins, on smaller fund managers (see Section 6.9.2 below). Whether these firms are able to continue operating in the UK will depend on the extent to which they can both pass on and absorb the cost of non-execution services, the latter of which is in turn dependent on their operating margin. Thus, we have further segmented the small segment according to margin¹³. We defined above average performers as those whose margin was above the FUM-weighted average margin (10%) for small fund managers (i.e. with FUM of less than £5bn) in 2002/03, and below average performers as those with a margin below this.

¹³ See Section 6.8 for an explanation of the derivation of 2002/03 margins.

4

MARKET SEGMENTATION

4.6 Hedge Funds

Responses to CP176 and interviews conducted with fund managers both highlighted ways in which customer relationships, level of charges, potential cost impacts, available strategies and their potential adoption differed for hedge funds compared to other fund managers. As little financial information exists for the majority of hedge-only funds, cost impacts have not been explicitly assessed for the segment, but hedge funds have been separately accounted for when assessing the availability and adoption of potential strategies in response to the Proposals.

4.7 Summary of our segmentation model

The resulting segmentation model of the fund management market contains three size segments (large, medium and small) and the following nine sub-segments:

- 1. Large, UK focused
- 2. Large, non-UK focused
- 3. Medium, UK focused
- 4. Medium, non-UK focused
- 5. Small, UK focused with above average operating performance
- 6. Small, non-UK focused with above average operating performance
- 7. Small, UK focused with below average operating performance
- 8. Small, non-UK focused with below average operating performance
- 9. Hedge Funds

The figures below show composition of the market according to our segmentation model on the bases of FUM and the number of firms in each segment.



4 MARKET SEGMENTATION



5. IMPLEMENTATION OF PROPOSAL II

5.1 Introduction

Proposal 2 of CP176 allows latitude for fund managers to:

- liaise with brokers to decide upon the most suitable method of acquiring nonexecution services; and
- decide on the most appropriate way of passing through the cost of these services to customers (i.e. through explicit service charges or an increase in the management fee).

This section summarises the views of fund managers and brokers on market outcomes that might result from these choices.

5.2 Method of acquiring non-execution services

Proposal 2 suggests that fund managers might unbundle the purchase of non-execution from execution services, or alternatively continue to buy bundled services and rebate the non-execution portion of these to their customers' funds.

We asked fund managers and brokers to predict the practices they are most likely to adopt if the proposal was implemented out of the following five options:

- A. Bundled execution and non-execution services, with fund managers rebating the non-execution element of these to customers.
- B. Separated execution and non-execution services, where non-execution services are purchased for a single bundled price.
- C. A menu of separately priced execution and non-execution services (e.g. a tariff pricing model).
- D. A mixture of the above depending on broker / customer.
- E. Some other arrangement.

5.2.1 Views expressed by fund managers

The table below shows the views expressed by fund managers in total and in each of the size segments introduced in the previous section. Half of the questionnaire respondents were willing to answer this question. Definitions of the large, medium and small segments are given in section 4 above.

5 IMPLEMENTATION OF PROPOSAL

Table 5.1 - Predicted method of acquiring execution services if Proposal 2 is implemented							
Option	Total	Large	Medium	Small			
A B C D E	15% 20% 34% 10% 21%	- 58% 8% 25% 8%	28% 7% 42% 5% 18%	- 50% - 50%			
Total	100%	100%	100%	100%			
	Source: Deloitte questionnaire						

The table shows that a wide variety of views prevailed but, on average, the most popular view was that brokers would implement a tariff pricing system allowing maximum flexibility over the purchase of non-execution services. It should be noted that fund managers choosing this option may have interpreted a tariff pricing model as applying to either cash or commission as a means of paying for these services.

The relative lack of appetite for option A (bundling and rebating) appears to stem primarily from a fear that cost recovery of rebated services will be low. This fear rests on the premise that customers will be less willing, through an increased management fee or explicit charges, to return the purchase price of non-execution services to fund managers once rebated to their funds, than to pay for these services in an unbundled environment.

Among those fund managers that chose option B (the most popular option among the large segment), a model of non-execution purchase arose in which the fund manager is provided with separate bundles of execution and non-execution services, absorbs the cost of the non-execution bundle, and raises management fees to the extent necessary to maintain margins. In this way, the level of services (principally research) consumed is fed into the management fee and customers can choose whether this fee is merited by the performance of the fund manager.

5.2.2 Views expressed by brokers

Brokers were not prescriptive on the method by which fund managers may receive services, indicating that solutions are likely to be fund manager driven and as such a series of options may be available, rather than the imposition of a uniform solution on the market. A general preference was expressed for the provision of sub-bundles paid for through commission rather than services purchased for cash, however cash purchase of services was not dismissed. It was envisaged that a menu of choices, ranging from execution-only to full service brokerage would be available to fund managers, as currently offered by many brokers.

5.3 Method of charging for non-execution services

5.3.1 Views expressed by fund managers

Fund managers were asked whether they anticipated recapturing the cost of non-execution services through explicit charges for each service or bundle of services, or whether management fees would be raised to cover extra costs. Of those willing to answer the question, 50% said they would seek to raise their management fee, 38% indicated they would need to negotiate with customers to establish whether costs could be passed through and the appropriate method of doing so, while 12% responded that they would not be able to pass costs through as customers would not be willing to bear these charges.

6. POTENTIAL INCREMENTAL COST IMPACT OF CP176

6.1 Introduction

A large number of responses received by the FSA from fund managers in response to CP176 maintained that incremental costs would be incurred as a result of implementation, and that these costs could not be passed through to customers through explicit charges or an increase in the management fee. The three reasons stated most frequently for a rise in costs were:

- though softed and bundled services would no longer be received as part of a commission payment to brokers, commissions would not fall by the full value of services lost. Though this would represent a loss of value to the fund manager's customer, rather than the fund manager, it would lessen the likelihood of fund managers being in a position to recover the costs from customers of direct purchase of these services on their behalf;
- the cost of non-execution services, particularly research, would rise when provided separately from the execution element of the bundle and purchased either as a subbundle, or separately for commission or cash; and
 - implementation of the Proposals would lead to substantial administration, Information Technology and customer relationship costs on an ongoing basis.

We look at the potential impact on each segment of the industry of these potential sources of incremental costs below, and examine the ability of each segment to pass them through to customers. To do this we look at the views expressed by fund managers from questionnaire responses and interviews. Where appropriate, we also note the opinion of brokers, trade associations and independent research companies.

Taking these views into account, we arrive at ranged conclusions on the potential incremental cost impacts on each market segment.

6.2 Impact of Proposal 1 on commission levels

This section assesses potential changes to commission levels resulting from the partial or total prohibition of softing.

6.2.1 Views expressed by fund managers

Implementation of Proposal 1 as drafted would prevent fund managers sourcing MPIS using soft commission. The proposal also invites views on restrictions on the softing of other services.

6 POTENTIAL INCREMENTAL COST IMPACT OF CP176

Fund managers that receive softed services were either pessimistic about their ability to recapture lost value of soft credits for customers through reduced commissions, or did not think the exercise worth the effort. On average, fund managers anticipated recovering 6% of the value of MPIS lost, and the same proportion of total softed services lost if a full ban was implemented. On a segmental basis, no large fund managers predicted that commissions attributable to MPIS would fall, while small and medium-sized fund managers forecast 7% and 4% decreases respectively. Weighting the analysis such that the opinions of those fund managers receiving the most services carry the most weight reduces the average decrease in commissions to 3% (value of MPIS) and 2% (value of total softed services).

6.2.2 Views expressed by brokers

Some support for fund managers' views that commissions would not fall if softing was partially or fully banned was provided by brokers. This was largely because, even amongst the largest brokers, the quantity of softing has fallen in recent years and so the shrinking or ending of the market for soft credits would not create an effect significant enough to alter overall commission levels.

6.2.3 Our conclusions

We broadly agree with the views expressed by fund managers and brokers that the value of services currently received by fund managers using soft credits will in many cases not be recaptured for customers through lower commissions. The following market dynamics support this view:

- in most cases, fund managers pay the same level of commissions for trades comprising a softed element as for those that do not;
- broker questionnaire responses indicate that the value of soft credits provided to fund managers is around 2.8% of total commissions received, a proportion lower than that indicated by fund managers in Section 3.2 above. A decrease in the value of credits provided from this low base level may not be enough to trigger a renegotiation of commission levels;
- smaller fund managers, including those utilising prime brokers, disproportionately benefit from softing and, with the exception of hedge funds trading at high volumes, are in a weaker position to negotiate lower commission levels than larger fund managers; and

6 POTENTIAL INCREMENTAL COST IMPACT OF CP176

many fund managers that currently receive softed services have little incentive to attempt to recover their value through lower commissions. Many larger fund managers have decreased their take-up of such services, possibly in part following the publication of CP176, and the value of services received represents a small part of their cost base. Additionally, some hedge funds trading at high volumes will have sufficient ability to either absorb extra costs incurred through the cash purchase of previously softed services, or greater flexibility over adjustments made to their management fee.

For these reasons, we believe the average commission decrease of between 2% and 6% depending on the basis of estimation used is reasonable, giving many fund managers limited incentive to attempt to recapture this amount from brokers in the absence of Proposal 2, and others limited ability to. However, if both proposals were implemented simultaneously, these levels of recapture may understate the incentive and ability to recapture commissions arising from the lost value of both softed and bundled services.

6.3 Impact of Proposal 2 on commission levels

This section assesses potential changes to commission levels resulting from the unbundling of the non-execution element of commissions. This may result from the adoption of a tariff pricing mechanism through which single services or groups of services are purchased for a cash price, or alternatively through the offer of sub-bundles for less than a full-service bundled commission rate.

For the purposes of this section, and following feedback from fund managers and brokers, non-execution services are defined broadly and may include each of the following services:

- trade advice;
- broker contact;
- proprietary or bespoke research (i.e. sourced from the executing broker or, under commission sharing arrangements, other brokers or independent research companies);
- non-proprietary research (i.e. that produced by brokers and sent to most fund managers – and thought by some to be marketing rather than research);
- conferences / seminars;
- company visits / introductions;
- equipment provided by the broker (e.g. online data feeds and other resources); and
- access to IPOs.
The above list is not exhaustive and covers various media including written material, telephone contact, e-mail, video conferencing and voicemail. As noted at Section 3.2, we have not considered the active working of a trade to be part of the bundle of non-execution services.

6.3.1 Views expressed by fund managers

Fund managers see more chance of proposal 2 resulting in lower commission rates than proposal 1. Large fund managers forecast on average that 38% of the value of bundled non-execution services would be regained through lower commissions. Medium-sized and small fund managers were less optimistic forecasting 15% and 37% recovery respectively, giving an average across all segments of 29% cost recovery. If responses are weighted by those receiving the most bundled services, this reduces to 21%

6.3.2 Views expressed by brokers

The majority of brokers interviewed stated that they already provide sub-bundles and execution-only services in addition to full-service brokerage. They indicated that the number of available sub-bundles or tariff pricing choices where services may be purchased for cash could increase, and as such fund managers would have greater flexibility than that already enjoyed to choose the range of services required for an appropriate commission rate. Therefore, if Proposal 2 was implemented, they predicted that close to 100% of the value of non-execution services previously included in the bundle would be recaptured by fund managers on their customers' behalf through lower commission rates.

6.3.3 Our conclusions

Recent market trends and activities undertaken by a significant proportion of brokers would appear to support their view that if the proposal was implemented, unbundled commission rates offered by traditional full-service brokers would compete with those offered by execution-only specialists, while a number of options would be available to fund managers up to and including a full-service offering. These trends and activities include:

Market trends

- falling commission levels in recent years are empirical evidence that, as an industry, fund managers exercise some market power over brokers (at least during a bear market), and may be in a position to use this to win reasonable service and commission levels if the Proposals are implemented;
- the continued growth of execution-only and programme trading, putting added emphasis on the execution component of commissions; and

increased awareness of the level of transaction costs among pension fund trustees and other customers following the Myners Report.¹⁴

Broker activities

- analysis undertaken to identify the cost of elements of a bundle offered to fund managers allowing for the separate pricing of components of the bundle;
- as an extension of this, construction of models to understand the profitability of each fund manager, and thus the break-even commission levels required from a set level of trades; and
- acceptance of and participation in commission sharing arrangements whereby a proportion of commissions are re-directed to other brokers or independent researchers for the provision of non-execution services.

In addition to the above, new Securities and Exchange Commission (SEC) rules¹⁵ in the United States have initiated reforms aimed at ending conflicts of interest which may have arisen partly from cross-subsidies between investment banking and research activities at large brokerage houses. Though the extent to which these have so far impacted the UK market is unclear, they may have given further impetus to the treatment of both sales and research teams as individual profit centres, and thus encouraged an internal requirement for each to be profitable.

In light of the above, it seems likely that, if Proposal 2 is enacted, commission levels would fall in line with the value of services removed from the bundle. However, this benefit may not fully accrue to some small and medium-sized fund managers that are relatively unprofitable for brokers on an individual basis due to a modest volume of trading and / or high uptake of added value services within the bundle currently received. These fund managers may not experience a commensurate fall in commissions for services lost, but as it is likely that bundled services will still be offered by brokers (see Section 5.2.2), they should have the option of continuing to receive these at a comparable commission rate to that currently paid,

¹⁴ "Institutional Investment in the United Kingdom: A Review", Paul Myners, March 2001.

¹⁵ Self Regulatory Organization (SRO) Research Analyst Rules announced by SEC Chairman William Donaldson, July 2003. Included:

Separation of Research Analyst Compensation from Investment Banking Influence. Investment banks are required to establish a compensation committee to review and approve the compensation of its research analysts that are primarily responsible for the preparation of the substance of research reports. The committee would report to the Board of Directors and may not have representation from the firm's investment banking department.

Pitch Meetings. Research analysts are insulated from investment banking interests by prohibiting them from participating in "pitches" or other communications for the purpose of soliciting investment banking business.

Prepublication Review of Research Reports. The pre-publication review and approval of research reports by persons not directly responsible for research is restricted. The rules also require that pre-publication communications about a research report between all non-research personnel and the research department be intermediated by legal or compliance staff.

not withstanding any additional costs faced due to a need to rebate the non-execution part of the bundle to customers.

Taking into account the above, we estimate conservatively that the part of commissions previously accounted for by non-execution services will fall on average by a range of 60%-80% if Proposal 2 is implemented. However, for many fund managers this decrease is likely to be 100%. This fall may be additionally supplemented by a continued increase in the takeup of execution-only and other forms of trading that do not require non-execution services, lowering the overall rate of commissions paid.

6.4 Impact of the Proposals on the value of non-execution services purchased

This section examines the possibility of changes in the value of non-execution services purchased if the Proposals are implemented. These may result from changes to either the price, quantity or quality of services received.

Among possible sources of increases in price are loss of economies of scope present in providing bundled services. Such economies primarily relate to the provision of advice on how a trade or series of trades may be undertaken. However, it is possible that economies also exist in the provision of non-trade advice delivered as a consensus view of the research, sales and trading teams familiar with the trading strategy of a fund manager.

Future increases in the price of non-execution services may also result from the following sources unrelated to the Proposals:

- ongoing restructuring of brokerage firms to prevent the cross-subsidisation of research departments from trading commissions (see Footnote 15 above);
- work carried out by brokers to assess the profitability of execution and other services received by individual fund managers; and
- evidence of a slowing down in the rate of decrease of commissions coupled with a partial recovery in stock markets which may in combination result in static or rising commissions in 2004/05.

If some of the above sources of increased price were supplemented by decreases in quantity or quality of research produced, some fund managers may have to spend more to find a suitable quality product in the sectors they wish to cover. However, offsetting the above are possible falls in the value of services received as a result of:

greater competition for research if independent research companies gain market share in types of research substitutable for outputs produced by brokers due to greater acceptance of cash payment (in an unbundled / tariff pricing environment) or the spread of commission sharing (under a bundling and rebating outcome); and/or

a fall in the quantity of non-execution services sought as a result of greater choice over those services received and those declined.

We asked fund managers and brokers to predict changes in the price and quality of nonexecution services received, and the quantity purchased / produced, following implementation of the Proposals. Generally fund managers were not willing to differentiate effects between services received. The comments below therefore relate to research unless otherwise stated.

6.4.1 Views expressed by fund managers

Softing

The cost of softed services should not rise as in the vast majority of cases these are provided to fund managers directly from service providers, therefore the price negotiation process will not change. This differs from the dominant system in the United States where brokers negotiate a bulk discount with service providers on behalf of fund managers. As noted at Section 3.4.1 above, a 4% decrease in the quantity of services purchased was predicted by fund managers.

Bundling

Half of the questionnaire respondents answered questions on the post-implementation levels of price, quality and quantity of research. Of these, a narrow majority predicted a fall in research prices irrespective of changes in their own demand such that, on average, the cost of research might be expected to fall by 8%. As noted at Section 3.4.1 above, if quantity forecasts are included, the change in the quantity-adjusted value of services purchased was predicted to fall by a range of 9% to 15%.

A narrow majority (56% versus 44%) believed that the quality of research would also deteriorate, as brokers continued to trim research departments and reduce coverage. Set against this, some fund managers thought that less non-proprietary written research would be produced, to which they attach a relatively low value, increasing the average quality of research received. No fund managers envisaged extra costs from undertaking a greater investment in in-house research.

6.4.2 Views expressed by brokers and independent researchers

Brokers

Brokers expressed a consensus view that prices for value-added alpha research services may rise for some or all fund managers for the following reasons:

- research and advice forwarded to fund managers was partly funded by the investment banking activities of large brokers but, in response to SEC actions noted at Section 6.3.3 above, this source of funding has diminished requiring full capture of research costs from fund managers; and
- research provided to a) smaller fund managers with a lower commission spend and b) those with the heaviest demand for broker input to validate the findings of inhouse research departments have in the past been subsidised by fund managers with a higher level of trades or less need of broker input to trading decisions. These cross-subsidies would end in an unbundled environment.

Brokers indicated that quantity of maintenance research (i.e. background research arising from company contact) which is useful as a supply of data for fund managers' financial models, may diminish in fringe sectors of the economy, at the extreme resulting in niche and small-cap companies sponsoring publication of research into themselves. A general fall in the quantity of research produced would depend on the success of fund managers' attempts to pass the cost through to customers. If a significant fall in demand is witnessed, a consequent fall in research provided to the UK market may result.

Brokers were sceptical that independent firms could fill any gap in research provision, particularly in the production of maintenance research requiring a high level of contact with companies.

Independent researchers

Independent researchers expressed the view that an unbundled environment would allow them to compete with brokers on a level playing field, providing the opportunity to attract star researchers from larger brokers. They commented that lower overheads currently allow research to be produced more cheaply at independent firms, thus gaining a greater market share as a result of more competitive conditions would decrease the aggregate market price of research.

6.4.3 Our conclusions

It is almost impossible to attempt to predict the future landscape of the research market, as this is affected by the scope of Proposals implemented by the FSA, strategies followed by market participants including fund managers' customers, and external factors including regulatory action in other territories and changes in research provision in line with the investment cycle. Notwithstanding these difficulties, and after challenging some of the opinions expressed above by fund managers and others, we think on a balance of probabilities basis that the following changes to the price, quality and quantity of research services might be expected:

- There is no reason to disbelieve fund managers' aggregate estimate of the price of research falling by 8% and the quantity-adjusted value of services received falling by 9%-15%. Similarly, fund managers' forecast 4% fall in the quantity of softed services purchased may result from a re-assessment of needs if Proposal 1 is implemented.
- However, this reduction in total price may hide increases falling most heavily on those medium and large fund managers that currently make the greatest use of labour-intensive analyst advice from multiple brokers, and small fund managers that have a high reliance on prime brokers for idea generation. Price discovery resulting from the separate charging of research may make the present level of consumption of these services unaffordable to some fund managers.
- Those currently receiving a low level of research within a bundled commission rate may experience an overall drop in the cost of research resulting from the ability to purchase the same amount of research for less commission or cash under or subbundle or tariff pricing arrangements. However, this option for some smaller fund managers may be limited by a lack of bargaining power with brokers over commission rates and form of services received (i.e. bundled / unbundled).
- Those making most use of research that is substitutable by the independent sector may experience price falls if this sector flourishes following implementation of the Proposals.
- Those fund managers that successfully circumvent the regulation by sourcing research from abroad (see Section 8.3 below) should escape any price increases in research provided to fully compliant companies. They will also be able to take advantage of price decreases in the UK market where these occur.
- There is no evidence to suggest that an overall fall in quality of research would result from implementing the Proposals.

It is beyond the scope of this report to attempt to judge the optimum level of research provision in the UK investment market, and whether this or another level of consumption may result from implementation of the Proposals.

6.5 Other costs

In this section, we summarise the views of fund managers that responded to our questionnaire on incremental costs that may be incurred following implementation of the Proposals, including:

- the cost of calculating appropriate price structures for presently bundled services;
- incremental costs associated with invoicing;
- costs associated with rebating (for those exercising this option); and
- incremental customer relationship costs.

As no respondents have undertaken a review to quantify the magnitude of these costs, we have not attempted to challenge estimates given in detail. However, the lack of industry consideration of these costs may hide the potential for a large bill for some fund managers.

In addition to the above categories of cost, some fund managers may need to re-negotiate cost structures with customers to avoid incurring a working capital cost resulting from the payment for non-execution services in advance of billing customers for those services. We understand that the majority of management fees are calculated and charged to funds on a daily basis, however for fund managers that levy charges to funds less often, and are unable to re-negotiate this practice, a carrying cost may accrue.

6.5.1 Calculation of unbundled services

This potential cost item is heavily contingent on the necessity for fund managers (as well as brokers) to participate in the process of calculating the payment mechanism for non-execution services, and the extent to which unbundling occurs in the market. The estimates below are therefore contingent on pricing mechanisms adopted in the market and are the high-end of a range that has a low end of zero.

50% of respondents were not in a position to estimate any incremental costs generated by this activity, one of these stating that this would be impossible until the scope of the Proposals was finalised, and another that it would depend on the trading models that brokers imposed on the market. Of the remainder, 22% believed no extra costs would be incurred while 78% thought that they would, however nearly half the respondents did not quantify these costs. Of those that forecast additional costs, these ranged from £5,000 to £300,000, much of which may be temporary. Four of 18 respondents stated that extra staff would need to be recruited, while four said that this would not be necessary.

One respondent stated that incremental resources would be required to scrutinise the quality of the non-execution services it received in order to determine whether they offer value for money. This would result in the recruitment of an extra analyst, at a fully loaded cost of £100,000 per annum. As it may be assumed that this function would only be undertaken if a net gain accrued from it, and that the ability to carry it out may be aided by the Proposals, we do not consider this to be an incremental cost.

6.5.2 Rebating costs

Only four of 18 respondents were willing to predict the likelihood (or not) of incremental costs arising from rebating, reflecting a general unwillingness to engage in this practice. Of these, three forecast that costs would be incurred ranging from £150,000 to £600,000 and comprised largely of one-off rather than ongoing costs. The fourth respondent did not believe costs would be incurred. Anecdotal evidence received suggests that the cost of implementing systems allowing rebating is likely to exceed that of modifying systems facilitating the receipt of unbundled services, however we have been unable to verify this.

6.5.3 Invoicing costs

Implementation of Proposal 2 may result in a change in invoicing systems and/or procedures for fund managers processing broker invoices or issuing invoices to customers. Of the 50% of respondents that were willing to speculate whether incremental invoicing costs would be incurred, 78% forecast incremental costs ranging from £5,000 to £1,000,000 (one-off costs) and £50,000 to £250,000 (ongoing costs), while 22% predicted no extra costs. Eight of 18 respondents forecast the recruitment of extra staff, while three thought recruitment would be unnecessary.

6.5.4 Customer relationship costs

Adoption of the Proposals may require fund managers to invest resources in informing customers about changes to fee and commission structures, and add a degree of complexity to customer relationships (e.g. extra negotiations around the level of non-execution services received). Of the 72% of respondents who expressed an opinion on this issue, 85% envisage incremental costs from increased customer relationship management. Of these, 30% expect a 'slight increase', and 55% a 'significant' impact, much of this ongoing rather than a one-off cost as arising from increased reporting to customers. However, only two estimates of extra costs were received, one being less than \pounds 10,000 and the other being a range of \pounds 0.7m - \pounds 1.4m.

6.5.5 Our conclusions

Assigning an average cost base impact for the industry or for individual segments arising from the cost categories considered in this Section may be spurious as:

- companies were generally unwilling to speculate on which costs may be incurred and have not carried out work to quantify these;
- the nature and magnitude of each type of cost will not become clear until the market has adjusted to the proposals that are implemented, which in turn depends on their final scope; and
- the cost categories are individual in their nature, and therefore no general impact is readily quantifiable.

In the light of these considerations, we have not included these costs in our conclusions on cost impacts at Section 6.9 below, however a general consideration as to their likely applicability to some companies in the small and medium-sized segments has been taken into account when arriving at assessments of strategies which may be followed by industry participants (see Section 8).

6.6 VAT implications of the Proposals

6.6.1 Introduction

At present commission charged by brokers is wholly exempt from VAT, so the cost for any fund manager receiving brokerage services, regardless of whether they have a taxable or VAT-exempt mandate, is the same. This is in contrast to the situation where non-execution services are provided separately from execution services (e.g. for the cash payment of independent research companies). These transactions are subject to VAT at the standard rate.

The VAT liability of a fund manager's fees depends upon the type of investment vehicle receiving the fund manager's services and in certain circumstances on whether the provider is an insurance company. Thus, a fund manager's fee may be subject to VAT at the standard rate of 17.5% (commonly referred to as taxable services), or exempt from VAT. For the avoidance of doubt any VAT incurred by the fund manager that relates solely to the provision of taxable services may be recovered by the fund manager. Conversely, VAT incurred by a fund manager that relates solely to the provision of VAT-exempt services may not be recovered. Therefore, for fund managers with VAT-exempt mandates, VAT incurred on bought-in services and goods represents an additional cost.

If non-execution services were unbundled and purchased either in cash or commission form, the potential exists for brokerage services to be newly constituted as two supplies, one of

taxable research, the other of VAT-exempt brokerage. This would create an additional VAT cost for fund managers which they would need to seek to recover from higher fees or explicit charges.

6.6.2 Precedent

The leading case on the VAT treatment of multiple services from the same supplier is the European Court of Justice decision in Card Protection Plan. This case identified several tests which must be applied when considering whether several services supplied by a single provider should carry the same VAT liability, or be treated separately for VAT purposes. We have paraphrased some of these tests below.

- 1. All of the circumstances of the case in question must be considered.
- 2. Every supply of a service must normally be regarded as distinct and independent.
- 3. Supplies which are a single supply from an economic point of view should not be split.
- If there is a supply where one element of it comprises the principal service, and the other elements are ancillary, this should be treated as a single supply of services. The VAT liability for the entire supply should follow that of the core supply.
- 5. A single price for the provision of services is not determinative of the issue.

The fifth test listed above would seem to favour the current treatment applied to brokerage services, as the separate identification of research would under this test simply be an administrative one rather than a change in the nature of services provided. However, the second test would appear to favour the view that there are two separate supplies, of brokerage and research.

There are other cases concerning this issue but none can provide definitive guidance on this matter. As can be implied by the tests given above, the treatment of brokerage services must be determined on a case by case basis. However, HM Customs & Excise (HMCE) may decide that a single policy on the provision of brokerage services should be agreed so that there is no uncertainty for businesses supplying such services.

6.6.3 Conclusion

It is not possible at this stage to determine what HMCE policy would be under such circumstances. The outcome of this may have some bearing on fund managers' attitude to compliance and other strategies as outlined in Section 8.

6.7 The ability of Fund Managers to recover incremental costs from customers

This section assesses the success with which fund managers may be able to recover incremental costs from their customers. These costs may arise from the need to:

- fund the purchase of at least some previously softed and/or bundled services, as detailed in Sections 6.2 and 6.3;
- meet any incremental costs arising from price increases to non-execution services, as described in Section 6.4;
- recover any incremental administration and systems costs, as outlined at Section 6.5; and
 - recover any liability for VAT costs, as summarised in Section 6.6.

6.7.1 Views expressed by fund managers

Softed services

Those questionnaire respondents that receive softed services predicted that only 14% of the cost of purchasing these under other arrangements following the implementation of Proposal 1 would be recoverable from customers. When this sample is weighted by those that receive the most services, the recovery rate drops to 7%, implying that some smaller fund managers are confident of a high level of recovery due to close customer relationships or a relatively small value of services received requiring recovery.

Bundled services

Fund managers were asked to project likely cost recovery from:

- non-execution services purchased separately;
- non-execution services purchased in a bundle; and
- administration, systems and VAT costs.

67% of respondents provided answers to one or more of these questions. Whether services are provided in a bundle made little difference to respondents who estimated on average that between 33% and 37% of costs could be recovered under either of these methods. Fewer respondents were willing to estimate cost recovery from administration and other costs, but those that did forecast an average 21% recovery rate. The relatively small size of this category of costs compared to ongoing execution costs also persuaded some respondents not to attempt to recapture the cost, and therefore predict no recapture. The table below summarises fund managers' views on potential cost recapture.

Table 6.1 - Fund managers' average estimated recapture of costs from customers											
Cost recovery from customers	Unbundled non- execution services weighted by importance in the bundle	Bundled non- execution services	Incremental administration, systems and VAT costs								
Percentage recapture through increased management fee or explicit charges	37%	33%	21%								
	Sou	rce: Deloitte fund man	agers questionnaire								

6.7.2 Our conclusions

Softed services

Fund managers' pessimism around the ability to recover costs related to MPIS, performance and valuation services, equipment and other technical services would appear justified as these are required to run the business in the same way that electricity and heating are necessary, and as a similarly predictable service, should logically be charged for in the same way. That they were previously received under a different method of purchase is unlikely to result in institutional customers agreeing to higher management fees for all but the best performing fund managers with the closest customer relationships.

The ability to recover the cost of softed research should not differ from that of bundled research, since customers' willingness to pay should depend on the service provided rather than the current means of purchase. We estimate below that the potential cost recovery of bundled research ranges from 50%-70%, therefore if research is assumed to constitute approximately 30% of softed services¹⁶, and 50%-70% is on average recoverable, total recovery of softed services may amount to between 15% and 21% of the value lost.

Bundled services

The key to fund managers' ability to recover costs lies in customers' willingness to both understand and accept a different method of charging in the UK compared to other territories. This will be influenced by several factors, including:

¹⁶ "An assessment of soft commission arrangements and bundled brokerage services in the UK", OXERA (2003)

- relative size and sophistication of customer, contributing to bargaining power this can be broadly divided between institutional and retail customers;
- location of customer (e.g. UK customers may be more likely to understand the Proposals);
- relationship between fund manager and customer;
- performance of the fund manager; and
- attitude of investment consultants (primarily those advising UK customers).

Each of the above will influence customers' attitudes towards costs (i.e. is the customer focused on the fund management fee or a more sophisticated expense ratio?) and transparency (retail customers may be less concerned than institutional customers). Those fund managers that have a high proportion of UK institutional customers may have the potential to recapture a greater proportion of costs as:

- these customers will be best positioned to understand the Proposals and have investment consultants who understand (and in many cases) support them; and
- they may treat transparency of costs as more of a priority than other types of customers.

These customers may also be happiest with an increase in the management fee as a predictable cost, rather than bearing fluctuating charges through individual charges for research costs.

The broadly positive stance of investment consultants is also likely to be important in fund managers' attempts to recover costs from UK packaged product retail customers. This class of customer may accept higher charges unless a widespread move to place funds with fund managers seeking to evade the regulation was made by members of their peer group.

Overseas customers may be least willing to accept new methods of charging, particularly if they are obliged to monitor new arrangements such as the rebating of non-execution costs to funds in tandem with the separate charging for these services. Set against this, the UK model may in time appear attractive to overseas institutional investors seeking similar levels of transparency to their UK counterparts if:

- the Proposals deliver a net saving to clients' funds (see Section 7 below) or, failing this, do not cause the Total Expense Ratio (TER) of UK funds to rise above that of overseas funds;
- post-implementation, UK customers and consultants perceive that they have benefited from greater transparency in the market; and/or

the proportion of execution-only and programme trades continues to increase, fostering a more global acceptance of the separate payment for additional nonexecution services.

On the first of these points, anecdotal evidence suggests that UK TERs are generally lower than those of overseas funds and therefore a small rise in the total expenses of funds would be unlikely to put UK funds at a disadvantage. As we have not been able to validate this assumption, additional analysis may need to be performed to do so.

As noted above, the ability to recapture costs from customers will be heavily influenced by the type of customer (institutional or retail) and their location (UK or overseas). Our assessment of customers' likely initial understanding and acceptance of the Proposals (i.e. before any savings or costs become apparent) is summarised in the diagram below. Full circles denote full understanding / acceptance, empty circles no understanding / acceptance.



The diagram shows that fund managers' customer bases will be key to their initial ability to recapture costs. As around 59% of institutional UK FUM is attributable to domestic customers¹⁷, a relatively high level of cost recovery should be possible from these customers. Some level of cost recovery should also be available from overseas customers, particularly for better performing fund managers that are able to maintain competitive TERs. On the basis of this analysis, fund managers' aggregate cost recovery estimates shown in Table 6.1 above may understate potential cost recovery. Though a general cost recovery forecast is not possible with a high degree of certainty, we would advance a range of 50-70% as a reasonable estimate, and we use this in our assessment of the total incremental cost impact on the fund management industry in Section 6.8, and of the potential impact on customers in Section 7.

¹⁷ Fund Management Survey, IMA (2002)

6.8 Potential incremental cost impact on the fund management industry

This section summarises the potential incremental cost impacts discussed above to arrive at a ranged estimated impact for each market segment. As noted at Section 6.5.5 above, we have not included incremental administration and systems costs below, though these contribute to our assessment of strategies available to each segment in Section 8. We have also excluded any VAT impact in our summary calculations as this is contingent on an HMCE assessment that is yet to be made. Data used includes:

- financial information extracted from the FSA database for 443 fund managers representing around 73% of the estimated size of the UK fund management market;
- data on the value of softed and bundled services currently received, and the value that may be purchased if the proposals are implemented, obtained from questionnaire respondents; and
- opinions of respondents on the proportion of the value of services that may be saved for customers through lower commissions¹⁸, and the proportion that may be recovered through increased management fees.

The financial data provided by the FSA has been used to calculate an approximate average margins impact on each segment of the market. As an element of depreciation is contained in the cost data, revenues less costs relates to what may be termed Earnings After Depreciation (EAD) rather than operating profit, and thus margins will understate operating margins.

The tables included in this section summarise the incremental cost impact on the total market and each market segment on four bases:

- Total impact impact on margins for the total market / segment when the total value of services lost, and the extent of cost recovery from customers, is applied to total revenues and costs.
- Average impact impact on margins for the average company in the market / segment when the average value of services lost and recovered from customers is applied to average revenues and costs.
- 3. Weighted average impact (financials and cost recovery) as at 2 above but where:
 - the value of services currently received is used to weight views on the value of services saved through lower commission, and the proportion of costs recovered from higher fees; and

¹⁸ Though lower commissions do not directly benefit fund managers' cost bases, they have an indirect impact as savings in commission rates may help to persuade customers that management fees should rise.

- revenues and costs are weighted by funds under management at each company.
- 4. Weighted average impact (cost recovery only) as at 2 above but incorporating the weighting of views on commission reductions / cost recovery as in the first sub-point at 3 above.

The purpose of applying these metrics to the views of questionnaire respondents¹⁹ and to our own conclusions²⁰ resulting from analysis and interviews is to arrive at an incremental cost impact range. The lowest value resulting from comparison of impacts using each metric forms the low-case in the impact ranges shown and the highest value the high-case. Ranged impacts inform the probability of each segment and sub-segment adopting the strategies outlined in Section 8 in response to the Proposals.

It should be noted that financial data included relates to a snapshot of industry performance in 2002/03²¹, the nadir of the bear market. Financial performance may be expected to improve as equity markets recover.

We have built into the tables a 4% decrease in the value of softed services received following implementation of the Proposals as forecast by respondents. We have also included a 12% decrease in the quantity-adjusted value of bundled services received. This is the mid-point of the ranged 9% to 15% forecast reduction predicted by respondents. Fund managers' forecasts relating to the post-implementation value of services purchased are discussed at Section 3.4.1 above.

6.8.1 Impact on margins predicted by fund managers: total market

The table below shows the impact on margins resulting from fund managers' predictions relating to changes in commission levels and cost recovery from customers for the total market using the metrics described above.

¹⁹ See Table 6.2, Table 6.4, Table 6.6, Table 6.8, Table 6.10 and Table 6.11 below

²⁰ See Table 6.3, Table 6.5, Table 6.7, Table 6.9, Table 6.12 and Table 6.13 below.

²¹ Financials relate to 2002 for some fund managers and 2003 for others, depending on year-end.

Table 6.2 - Marg	gins impa	ct foreca	st by fund	l manager	s: Total r	narket				
Cost impact of:	Margin 2002/03		Margin 2002/03 Margin 2002/03 Proportion of service value recaptured through lower commissions LOW		Propo servic recove cust	ortion of e costs red from omers	Adjuste 200	d margin)2/03	Difference (percentage points)	
	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH
Partial ban on softing	13.47%	13.85%	2.09%	3.57%	6.78%	13.75%	12.57%	13.08%	(0.90%)	(0.77%)
Total ban on softing	13.47%	13.85%	1.95%	3.57%	2.94%	14.25%	10.19%	12.06%	(3.27%)	(1.79%)
Ban on bundling	13.47%	13.85%	16.90%	28.77%	5.32%	34.44%	4.98%	7.96%	(8.48%)	(5.89%)
CP176 as drafted	13.47%	13.85%					3.94%	7.13%	(9.53%)	(6.72%)
Total ban on both softing and bundling	13.47%	13.85%					1.56%	6.17%	(11.90%)	(7.68%)
				Source:	FSA data	abase / De	loitte fun	d manage	ers' quest	ionnaire

Fund managers' forecasts would result in a fall in margin from 13%-14% to a range of 4%-7% if the Proposals were implemented as drafted. If softing was banned entirely, margins fall further to a range of 2%-6%.

6.8.2 Deloitte analysis of potential margins impact: Total market

Γ

As summarised at Sections 6.2 and 6.3 above, we share fund managers' views regarding their limited ability to secure falls in commission resulting from the loss of softed services, but believe commissions accounted for by bundled services will decrease on average by an estimated range of 60%-80% as a result of the loss of these services.

At section 6.7, we concluded that depending on customer base and performance, fund managers on average may be able to recapture 50%-70% of the cost of bundled services from customers through a higher management fee, and between 15% and 21% of the cost of softed services, but a lower proportion of the cost of MPIS.

Using these assumptions, the Table below summarises our estimate of the impact of increased costs on margins.

Table 6.3 - Deloitte analysis of potential margins impact: Total market												
Cost impact of: Margin 2002/03		Proportion of service value recaptured through lower commissions		Propo servic recover custe	rtion of e costs red from omers	Adjuste 200	d margin 02/03	Difference (percentage points)				
	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH		
Partial ban on softing	13.47%	13.85%	2.09%	3.57%	6.78%	13.75%	12.57%	13.08%	(0.90%)	(0.77%)		
Total ban on softing	13.47%	13.85%	1.95%	3.57%	15.00%	21.00%	10.65%	12.39%	(2.82%)	(1.46%)		
Ban on bundling	13.47%	13.85%	60.00%	80.00%	50.00%	70.00%	8.99%	11.24%	(4.48%)	(2.61%)		
CP176 as drafted	13.47%	13.85%					8.01%	10.47%	(5.45%)	(3.38%)		
Total ban on both softing and bundling	13.47%	13.85%					6.09%	9.78%	(7.37%)	(4.07%)		
						Sour	ce: FSA d	latabase /	Deloitte a	analysis		

Our analysis shows an estimated fall in margin from 13%-14% to a range of 8%-10% if the Proposals were implemented as drafted. If softing was banned entirely, margins fall to a range of 6%-10%.

6.8.3 Impact on margins predicted by fund managers: Large fund manager segment

The table below shows the impact on margins for the large fund manager market segment as predicted by respondents in this segment.

Table 6.4 - Margins impact forecast by fund managers: Large fund managers												
Cost impact of:	Margin 2002/03		Proportion of service value recaptured through lower commissions		Proportion of service costs recovered from customers		Adjusted margin 2002/03		Difference (percentage points)			
	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH		
Partial ban on softing	12.30%	12.31%	0.00%	0.00%	0.00%	7.08%	11.68%	11.98%	(0.62%)	(0.33%)		
Total ban on softing	12.30%	12.31%	0.00%	0.00%	0.00%	7.01%	10.96%	11.73%	(1.34%)	(0.58%)		
Ban on bundling	12.30%	12.31%	33.71%	38.25%	43.41%	60.00%	6.67%	8.82%	(5.63%)	(3.49%)		
CP176 as drafted	12.30%	12.31%					6.09%	8.28%	(6.21%)	(4.03%)		
Total ban on both softing and bundling	12.30%	12.31%					5.33%	7.55%	(6.97%)	(4.75%)		
				Source:	FSA data	base / De	loitte fun	d manage	ers' quest	ionnaire		

Fund managers forecasts would result in a fall in margin from 12% to a range of 6%-8% if the Proposals were implemented as drafted. If softing was banned entirely, margins fall to a range of 5%-8%.

6.8.4 Deloitte analysis of potential margins impact: Large fund managers

Using the assumptions detailed at Section 6.8.2 above, the Table below summarises our estimate of the impact of increased costs on margins for large fund managers.

Table 6.5 - Deloi	itte analy	sis of pot	ential ma	rgins imp	act: Larg	e fund ma	inagers			
Cost impact of:	Margin 2002/03		Proportion of service value recaptured through lower commissions		Proportion of service costs recovered from customers		Adjusted margin 2002/03		Difference (percentage points)	
	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH
Partial ban on softing	12.30%	12.31%	0.00%	0.00%	0.00%	7.08%	11.68%	11.98%	(0.62%)	(0.33%)
Total ban on softing	12.30%	12.31%	0.00%	0.00%	15.00%	21.00%	11.07%	11.81%	(1.23%)	(0.49%)
Ban on bundling	12.30%	12.31%	60.00%	80.00%	50.00%	70.00%	7.33%	9.80%	(4.97%)	(2.51%)
CP176 as drafted	12.30%	12.31%					6.71%	9.47%	(5.60%)	(2.84%)
Total ban on both softing and bundling	12.30%	12.31%					6.10%	9.30%	(6.20%)	(3.01%)
						Sour	ce: FSA d	latabase /	Deloitte a	analysis

Our analysis shows an estimated fall in margin from 12% to a range of 7%-9% if the Proposals were implemented as drafted. If softing was banned entirely, margins fall to a range of 6%-9%.

6.8.5 Margins impact forecast by fund managers: Medium-sized fund manager segment

The table below shows the impact on margins for the medium-sized fund manager market segment as predicted by respondents in this segment.

Table 6.6 - Margins impact forecast by fund managers: Medium-sized fund manager												
Cost impact of:	Margi	n 2002/03	Propo servic recapture	rtion of e value ed through	Propo servic recover	rtion of e costs red from	Adjuste 200	d margin 2/03	Difference (percentage points)			
	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH		
Partial ban on softing	15.17%	17.32%	4.17%	7.44%	2.50%	7.13%	14.10%	16.15%	(1.07%)	(1.17%)		
Total ban on softing	15.17%	17.32%	4.17%	6.95%	3.50%	9.24%	13.59%	15.62%	(1.58%)	(1.70%)		
Ban on bundling	15.17%	17.32%	14.55%	20.25%	22.50%	25.61%	11.37%	13.14%	(3.80%)	(4.18%)		
CP176 as drafted	15.17%	17.32%					10.29%	11.97%	(4.88%)	(5.35%)		
Total ban on both softing and bundling	15.17%	17.32%					9.79%	11.44%	(5.38%)	(5.88%)		
				Source:	FSA data	base / De	loitte fun	d manage	ers' questi	ionnaire		

Fund managers forecasts would result in a fall in margin from 15%-17% to a range of 10%-12% if the Proposals were implemented as drafted. If softing was banned entirely, margins fall to a range of 10%-11%.

6.8.6 Deloitte analysis of potential margins impact: Medium-sized fund managers

The Table below summarises our estimate of the impact of increased costs on margins for medium-sized fund managers.

Table 6.7 - Deloitte analysis of potential margins impact: Medium-sized fund managers													
Cost impact of:	Margin 2002/03		Proportion of service value recaptured through lower commissions		Proportion of service costs recovered from customers		Adjusted margin 2002/03		Difference (percentage points)				
	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH			
Partial ban on softing	15.17%	17.32%	4.17%	7.44%	2.50%	7.13%	14.10%	16.15%	(1.07%)	(1.17%)			
Total ban on softing	15.17%	17.32%	4.17%	6.95%	15.00%	21.00%	13.78%	15.84%	(1.39%)	(1.48%)			
Ban on bundling	15.17%	17.32%	60.00%	80.00%	50.000%	70.000%	12.72%	15.63%	(2.45%)	(1.69%)			
CP176 as drafted	15.17%	17.32%					11.64%	14.46%	(3.53%)	(2.86%)			
Total ban on both softing and bundling	15.17%	17.32%					11.33%	14.15%	(3.85%)	(3.17%)			
						Sour	ce: FSA d	atabase /	Deloitte a	analysis			

Our analysis shows an estimated fall in margin from 15%-17% to a range of 12%-14% if the Proposals were implemented as drafted. If softing was banned entirely, estimated margins fall to a range of 11%-14%.

6.8.7 Margins impact forecast by fund managers: Small fund managers

The table below shows the impact on margins for the small fund manager market segment as predicted by respondents in this segment.

Table 6.8 - Margins impact forecast by fund managers: Small fund managers												
Cost impact of:	Margin 2002/03		Proportion of service value recaptured through lower commissions		Proportion of service costs recovered from customers		Adjusted margin 2002/03		Difference (percentage points)			
	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH		
Partial ban on softing	10.14%	13.90%	1.48%	6.94%	6.73%	33.33%	8.19%	13.00%	(1.95%)	(0.90%)		
Total ban on softing	10.14%	13.90%	1.38%	6.94%	2.03%	33.33%	3.42%	10.59%	(6.72%)	(3.31%)		
Ban on bundling	10.14%	13.90%	16.00%	36.67%	1.69%	33.33%	(1.96%)	5.65%	(12.10%)	(8.25%)		
CP176 as drafted	10.14%	13.90%					(3.91%)	4.25%	(14.04%)	(9.65%)		
Total ban on both softing and bundling	10.14%	13.90%					(8.68%)	1.05%	(18.82%)	(12.85%)		
				Source:	FSA data	ibase / De	loitte fun	d manage	ers' quest	ionnaire		

Fund managers forecasts would result in a fall in margin from 10%-14% to a range of -4% to 4% if the Proposals were implemented as drafted. If softing was banned entirely, estimated margins fall further to -9% to 1%.

6.8.8 Deloitte analysis of potential margins impact: Small fund managers

The table below summarises our estimate of the impact of increased costs on margins for small fund managers.

Table 6.9 - Deloitte analysis of potential margins impact: Small fund managers												
Cost impact of:	Margin 2002/03		Proportion of service value recaptured through lower commissions		Proportion of service costs recovered from customers		Adjusted margin 2002/03		Difference (percentage points)			
	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH	LOW	HIGH		
Partial ban on softing*	10.14%	13.90%	1.48%	6.94%	6.73%	33.33%	8.19%	13.00%	(1.95%)	(0.90%)		
Total ban on softing	10.14%	13.90%	1.38%	6.94%	15.00%	21.00%	4.31%	11.23%	(5.83%)	(2.67%)		
Ban on bundling**	10.14%	13.90%	60.00%	80.00%	50.00%	70.00%	3.98%	10.37%	(6.15%)	(3.53%)		
CP176 as drafted	10.14%	13.90%					2.04%	9.47%	(8.10%)	(4.43%)		
Total ban on both softing and bundling	10.14%	13.90%					(1.85%)	7.70%	(11.98%)	(6.20%)		
						Sour	ce: FSA d	atabase	/ Deloitte a	analysis		

Our analysis shows an estimated fall in margin from 10% - 14% to a range of 2%-9% if the Proposals were implemented as drafted. If softing was banned entirely, estimated margins fall further to -2% to 8%.

6.9 Summary of potential incremental cost impact

In this section we summarise the potential cost impacts highlighted in Section 6.8. This summary relies on the assumption that every company has a static operating model and

therefore would not be able to mitigate cost impacts (for example by adopting cost-cutting strategies).

6.9.1 Fund managers' views

Based on the analysis in the previous section, fund managers' conclusions on the potential incremental cost impact for each segment and sub-segment of the market are summarised in the table below.

Table 6.1	10 - Summary of	fund managers' for	recasts of p	otential marg	ins impact		
Impact o	on	Margin	2002/03	Adjuste 200	d margin)2/03	Diffe (percenta	erence age points)
		LOW	HIGH	LOW	HIGH	LOW	HIGH
Total ma	arket	13.47%	13.85%	3.94%	7.13%	(9.53%)	(6.72%)
Large		12.30%	12.31%	6.09%	8.28%	(6.21%)	(4.03%)
Large	UK	5.03%	6.06%	(4.09%)	(1.68%)	(9.12%)	(7.74%)
Large	non-UK	16.67%	17.03%	13.40%	14.25%	(3.27%)	(2.79%)
Medium		15.17%	17.32%	10.29%	11.97%	(4.88%)	(5.35%)
Mediu	m UK	21.17%	22.77%	16.27%	17.52%	(4.89%)	(5.25%)
Mediu	m non-UK	12.60%	15.04%	7.73%	9.64%	(4.87%)	(5.40%)
Small		10.14%	13.90%	(3.91%)	4.25%	(14.04%)	(9.65%)
Small	UK1	43.22%	46.08%	25.37%	34.82%	(17.85%)	(11.26%)
Small	UK2	(7.00%)	(6.94%)	(28.59%)	(19.88%)	(21.59%)	(12.94%)
Small	non-UK1	42.80%	45.84%	15.48%	33.44%	(27.32%)	(12.40%)
Small	non-UK2	(2.14%)	(0.98%)	(12.15%)	(8.98%)	(10.01%)	(8.00%)
Key:	Small 1 = Abo	ve average margins	2002/03				
	Small 2 = Belo	w average margins	2002/03				
		:	Source: FSA	database / D	eloitte fund	managers' qu	uestionnaire

According to fund managers' forecasts, industry margins could fall by between 7 and 9 percentage points with the small segment suffering the largest potential fall of 10 to 14 percentage points.

Based on these views, we calculated pre- and post-implementation margins for each of the 443 companies contained in the FSA fund managers' database, assuming the Proposals are implemented as drafted (i.e. stopping short of a complete ban on softing). The table below shows the impact on different industry segments in terms of the number of firms moving from a positive to a negative margin post-implementation.

	No. of firms	% of total FUM	Firms with positive margin 02/03	% of segment FUM	Firms with negative margin 02/03	% of segment FUM	Firms whose margin may have become negative because of CP176	% of segment FUM	% of firms that may have experienced negative margin following implementation of CP176 (as % of segment FUM)
Total	443	100%	327	79%	116	21%	184	22%	43%
By segment	=		·						
Large	11	46%	9	79%	2	21%	3	30%	50%
Large UK	6	29%	4	67%	2	33%	2	27%	60%
LargeNon- UK	5	17%	5	100%	0	0%	1	18%	18%
Medium	48	42%	35	81%	13	19%	7	16%	35%
Medium UK	14	13%	12	94%	2	6%	1	9%	15%
MediumNon UK	- 34	29%	23	76%	11	24%	6	15%	40%
Small	384	13%	283	72%	101	28%	174	64%	92%
SmallUK1 SmallUK2	103 104	3% 3%	103 62	100% 63%	0 42	0% 37%	38 58	37% 57%	37% 94%
SmallNon- UK1	66	2%	66	100%	0	0%	30	53%	53%
SmallNon- UK2	111	4%	52	44%	59	56%	48	43%	99%
Key:	Small 1 =	Above av	verage marg	ins 2002/0	3				
:	Small 2 =	Below av	erage marg	ins 2002/03	3				
				Source	e: FSA Datal	base / Delo	itte fund m	nanagers'	questionnaire

 Table 6.11 - Analysis of positive and negative margins as forecast by fund managers

In 2002/03, 116 firms representing 21% of market FUM had negative margins. Respondents' expectations indicate that, if the Proposals are implemented, a further 184 firms (22% of market FUM) would have experienced negative margins on this basis.

Of small above average performing companies (sub-segments SmallUK1 and SmallNon-UK1), where no firms currently show a negative margin, companies representing 37% and 53% of FUM would have reported negative margins according to fund managers' expectations. Amongst the below average performance sub-segments (SmallUK2 and SmallNon-UK2), 94% and 99% of companies would have experienced negative margins.

6.9.2 Summary of Deloitte analysis

Based on our analysis and views regarding changes to commission levels and recovery of incremental costs from customers, our estimates of the potential incremental cost impact for each market segment are summarised in the table below.

Table 6.1	2 - Summary of D	eloitte analysis of p	otential marg	ins impact:			
Impact o	n	Margin	2002/03	Adjuste 200	d margin)2/03	Diffe (percenta	rence lae points)
		LOW	HIGH	LOW	HIGH	LOW	HIGH
Total ma	rket	13.47%	13.85%	8.01%	10.47%	(5.45%)	(3.38%)
Large	IK	12.30%	12.31%	6.71% (3.14%)	9.47%	(5.60%) (8.17%)	(2.84%)
Large r	ion-UK	16.67%	17.03%	13.72%	15.18%	(2.95%)	(1.85%)
Medium Mediun Mediun	n UK n non-UK	15.17% 21.17% 12.60%	17.32% 22.77% 15.04%	11.64% 17.63% 9.08%	14.46% 19.97% 12.16%	(3.53%) (3.54%) (3.52%)	(2.86%) (2.80%) (2.88%)
Small Small U Small U Small r Small r	JK1 JK2 Ion-UK1 Ion-UK2	10.14% 43.22% (7.00%) 42.80% (2.14%)	13.90% 46.08% (6.94%) 45.84% (0.98%)	2.04% 32.93% (19.45%) 27.04% (7.91%)	9.47% 40.85% (12.69%) 40.09% (4.29%)	(8.10%) (10.30%) (12.45%) (15.76%) (5.77%)	(4.43%) (5.22%) (5.75%) (5.75%) (3.32%)
Key:	Small 1 = Abov	ve average margins	2002/03				
	Small 2 = Belo	w average margins	2002/03				
				Sou	irce: FSA dat	abase / Deloi	itte analysis

Margins for the industry as a whole are estimated to fall by between 3 and 5 percentage points. The least impact is expected on the medium segment, and the most on the small segment, which may see falls in margin of between 5 and 8 percentage points.

This summary of our estimates can also be converted into an analysis of positive and negative margins on the same basis as in Table 6.11, as shown in the table below.

	No. of firms	% of total FUM	Firms with positive margin 02/03	% of segmen t FUM	Firms with negative margin 02/03	% of segment FUM	Firm margii becon becaus	ns wh n may ne ne se of	ose / have gative CP176	% of segment FUM			% of firms that may have experienced negative margin following implementation of CP176 (as % of segment FUM)		
Total	443	100%	327	79%	116	21%	Worst 107	-	Best 83	Worst 16%	-	Best 5%	Worst 37%	-	Best 26%
By segment								=			=			=	
Large	11	46%	9	79%	2	21%	2	-	0	21%	-	0%	42%	-	21%
Large UK	6	29%	4	67%	2	33%	2	-	0	27%	-	0%	60%	-	33%
Large Non- UK	5	17%	5	100%	0	0%	0	-	0	0%	-	0%	0%	-	0%
Medium	48	42%	35	81%	13	19%	5	-	3	14%	-	5%	33%	-	24%
Medium UK	14	13%	12	94%	2	6%	1	-	1	9%	-	9%	15%	-	15%
Medium Non- UK	34	29%	23	76%	11	24%	4	-	2	12%	-	2%	36%	-	27%
Small	384	13%	283	72%	101	28%	100	-	80	42%	-	32%	70%	-	60%
SmallUK1	103	3%	103	100%	0	0%	5	-	3	9%	-	7%	9%	-	7%
SmallNon-	104	3%	62	100%	42	0%	40	-	30 7	49%	-	39% 110/	00%	-	10%
UK1 SmallNon-	00	2 70	00	100 %	0	0 %	11	-	1	23%	-	1170	23%	-	1170
uk2	111	4%	52	44%	59	56%	39	-	32	37%	-	29%	94%	-	86%
Key:	Small	1 = Abo	ve averaç	ge margins	s 2002/03										
	Small	2 = Belo	ow averag	e margins	\$ 2002/03										
								:	Source	: FSA	Data	abase /	Deloitte	e an	alysis

Table 6.13 - Analysis of positive and negative margins based on Deloitte cost impact assessment

According to this analysis, of the 443 companies included in the FSA's database, a range of 83 to 107 out of 327 companies with a positive margin in 2002/03 may have been vulnerable to reporting a negative margin as a result of implementation of the Proposals at that snapshot in time (equal to 5% - 16% of total FUM), taking the number of firms with a negative margin up to a range of 199 to 223 (equal to 26% - 37% of total FUM).

7. NET IMPACT ON CUSTOMERS

7.1 Introduction

Having examined the potential for incremental costs to be incurred by fund managers in the previous section, we assess below the resulting aggregate impact on customers of UK-regulated fund managers that fully comply with the FSA's Proposals (i.e. before consideration of evasion strategies).

The impact on customers is analysed after accounting for the following three factors:

- decreases to commission levels;
- changes to the size of the market for MPIS and bundled services; and
- fund managers' ability to recover incremental costs from customers through increased management fees.

The analysis is indicative of the financial impact on customers' funds that may result from these factors. It is not intended to account for other types of impacts that may affect some but not all customers including:

- potential increased tax liability for some investment vehicles where management fees increase (see Section 6.6.1 above);
- changes to fund managers' trading behaviour or strategy; and
- changes in market structure (e.g. consolidation).

This section does not therefore provide a complete assessment of the way customers may be impacted if the Proposals are implemented.

In Section 7.2 we assess the potential net customer impact using fund managers' predictions in relation to the three factors listed above. In Section 7.3, we apply our own assessment of the potential impact of changed commission levels, purchasing patterns and cost recovery to arrive at an alternative analysis of potential customer impact.

7.2 Analysis of net customer impact using fund managers' views

The table below uses responses to the fund managers questionnaire to estimate the net impact on customers if the Proposals are implemented. Fund managers' views are shown on a total market basis (e.g. the changed value of services purchased and the extent of cost recovery from customers are applied to total revenues and costs). Best and worst case scenarios are estimated from the point of view of the customer. Thus, a large decrease in commissions paid, large fall in fund managers' demand for non-execution services and low

levels of cost recovery are shown as best case scenarios for customers while the reverse would be worst case scenarios.

Table 7.1 - Net customer impact resulting from respondents' views						
		£m / %				
Decrease in commission levels	Worst case		Best case			
Total UK and international commissions 2003	2,878	-	3,435			
Commission attributable to MPIS	58.2	-	69.5			
Estimated decrease in commissions - MPIS		2%				
Commission attributable to bundled services	653.1	-	779.6			
Estimated decrease in commissions – bundled services	10	29%	4.4			
Decrease in commissions: MPIS	1.2	-	1.4			
Total decrease in commission payments	100.4	-	222.0			
Decrease in quantity/price of services purchased	Worst case	-	Best case			
	monot ouco		2001 0400			
Value of MPIS purchased	49.7	-	59.4			
Fund manager % saving through lower value of MPIS purchased		4%				
Fund manager % saving through lower value of bundled	9%	-	15%			
Fund manager saving on MPIS	2.0	_	24			
Fund manager saving on hundled services	58.9	-	114 0			
Post-implementation market size – MPIS	47.7	-	57.0			
Post-implementation market size – bundled services	594.2	-	665.6			
Estimated cost recovery from customers	Worst case		Best case			
% cost recovery - MPIS		7%				
% cost recovery - bundled services		32%				
Cost recovery - MPIS	3.2	-	3.9			
Cost recovery - bundled services	192.2	-	215.3			
Total cost recovery	195.4		219.1			
Aggregate customer impact	Worst case		Best case			
Aggregate customer impact: MPIS	2.0	-	2.4			
Aggregate customer impact: bundled services	5.8	-	(7.2)			
Total customer impact	7.8	-	(4.8)			
Source: FSA database / Deloitte fund managers questionnaire						

The table indicates that, if fund managers' views on the variables above came to fruition, funds held by customers managed by UK-regulated fund managers may see an impact ranging from a net saving of £4.8m to a net cost of £7.8m per annum. Whereas fund managers' predictions always result in a net cost resulting from Proposal 1, a saving or cost may result from Proposal 2.

If softing was to be banned altogether, the higher value of services received is offset by slightly different forecasts regarding changes to commission levels and cost recovery. These factors produce a broadly similar customer impact ranging from a $\pounds 6.3m$ net loss to a $\pounds 6.6m$ net gain.

7.3 Analysis of net customer impact from our assessment of the market

In Section 6, we outlined that fund managers' views appeared reasonable that:

commissions would not fall significantly to compensate for the lost value of softed services, and that this value was unlikely to be recovered from customers; and

7 **NET IMPACT ON CUSTOMERS**

the value of non-execution services purchased may fall by around 4% (services currently softed) and between 9% and 15% (bundled services).

However, we diverged from the views expressed by respondents on:

- the potential decrease in commissions resulting from Proposal 2, as we share brokers' views that any residual commission charged over and above executing and working trades should be more than 50% lower than commissions previously charged for bundled services. We have estimated a conservative ranged reduction of 60%-80% in this component of commissions (see Section 6.3.3); and
- the potential for recovery of the cost of bundled services and the research element of softed services from customers, which we have estimated to be in the region of 50%-70% (see Section 6.7.2).

If these views are applied to the analysis above (i.e. fund managers' assumptions relating to bundled services are challenged but those relating to softing are accepted), the estimated impact on customers' funds would range from a £26.1m net loss to a £288.4m net gain per annum. This is shown in the Table below.

Table 7.2 - Net customer impact resulting from Deloitte market analysis						
		£m / %				
Decrease in commission levels	Worst case		Best case			
Total UK and international commissions 2003	2.878	-	3.435			
Commission attributable to MPIS	58.2	-	69.5			
Estimated decrease in commissions - MPIS		2%				
Commission attributable to bundled services	653.1	-	779.6			
Estimated decrease in commissions – bundled services	60%	-	80%			
Decrease in commissions: MPIS	1.2	-	1.4			
Decrease in commissions: bundled services	391.9	-	623.7			
Total decrease in commission payments	393.1	-	625.1			
Decrease in quantity/price of services purchased	Worst case		Best case			
Value of MPIS purchased	49.7	-	59.4			
Fund manager % saving through lower value of MPIS purchased		4%				
Fund manager % saving through lower value of bundled services purchased	9%	-	15%			
Fund manager saving on MPIS	2.0	-	2.4			
Fund manager saving on bundled services	58.9	-	114.0			
Post-implementation market size – MPIS	47.7	-	57.0			
Post-implementation market size – bundled services	594.2	-	665.6			
Estimated cost recovery from customers	Worst case		Best case			
% cost recovery - MPIS		7%				
% cost recovery - bundled services	70%	-	50%			
Cost recovery – MPIS	3.2	-	3.9			
Cost recovery – bundled services	416.0	-	332.8			
Total cost recovery	419.2		336.7			
Aggregate customer impact	Worst case		Best case			
Aggregate customer impact: MPIS	2.0	-	2.4			
Aggregate customer impact: bundled services	24.1	-	(290.9)			
Total customer impact	26.1	-	(288.4)			
	Source: FSA database / Deloitte analysis					

Our acceptance of fund managers' views on the impact of softing results in the same small negative impact from Proposal 1 as shown in Table 7.1. However, this relies on

7 NET IMPACT ON CUSTOMERS

respondents' predictions that a slightly higher amount of cost recovery may be available from customers than the level of falls in commission available from brokers. This trade-off between commissions and cost recovery could easily swing in the opposite direction, and may also be affected by decisions to implement the Proposals simultaneously or individually, as fund managers may be more able to recover the entire value of non-execution services foregone through lower commissions, than to recover the value of softed services alone.

The indication that Proposal 2 might result in a net saving rather than a net cost to customers results from:

- a smaller value of currently bundled non-execution services to be passed through to customers resulting from decreased demand; and
- the estimated fall in commissions attributable to these services exceeding that of the level of cost recovery from customers. However, even if it was assumed that a 50%-70% fall in commissions occurred, matching the assumed level of cost recovery, the results would still range from a £91m net loss to a £210m net saving per annum.

If the analysis is extended to all softed services, an impact range of a £44.9m net loss to a $\pounds 273.2m$ net gain per annum is estimated. This is because a 50%-70% rate of cost recovery from customers is assumed for the research portion of softed services (estimated at 30%) as described at Section 6.7.2 above. This results in a ranged recovery of the cost of softed services of between 17% and 23%, far exceeding the reduction in commission attributable to these services if softing was banned which, in line with fund managers' views, is estimated at 2%.

If net savings did accrue to customers' funds following implementation of the Proposals, fund managers adopting evasion strategies (see Section 8) may suffer some customer loss as the performance of their funds is disadvantaged. A positive customer impact may also encourage overseas customers to transfer assets to UK-regulated fund managers.

8.1 Introduction

We have sought to identify ways in which different fund managers might respond to the implementation of the Proposals, with particular reference to the costs that they may incur as detailed in Section 6. The strategies outlined below have been derived from interviews with fund managers and brokers, responses to questionnaires and CP176 consultation responses. They are, by definition, hypothetical since no equivalent 'shock' to the market has previously been introduced that might act as a benchmark.²² Similarly, no other territory has yet introduced the changes proposed by the FSA.

We have tried to analyse below each of the courses of action that fund managers may be incentivised to take. In each case we have explained how we have defined the strategy, assessed the advantages and disadvantages of following it, and as a result, estimated the proportion of each market segment that may do so (see section 4 for the definition of these segments). Our assessment is summarised at Table 8.3. and is based on an assumption that fund managers will act rationally and make decisions based on their economic interest in response to the Proposals.

We have excluded from our analysis those strategies which we think are unlikely to be pursued. These are summarised below:

Re-domiciling funds

Following discussion with the FSA, we have worked on the assumption that if the Proposals are implemented, they would account for all activities of UK authorised fund managers. With the exception of the sourcing of non-execution services from overseas, circumvention would require the relocation of these activities. Therefore, it is assumed that that the movement of funds to new locations (e.g. Dublin or Luxembourg) would not bypass the Proposals.

Net trading

A number of respondents to CP176 remarked upon the potential for equity trading to move to a net basis such that payment for execution and other services takes place through the spread between bid and sell prices, making commissions (and the Proposals in relation to them) redundant. This would bring the method of payment for UK equity trade execution in line with that for fixed income products.

²² Research was carried out on regulatory changes and cost shocks to the UK (e.g. big bang) and other markets, but we concluded that little similarity with the Proposals was apparent in relation to potential changes to pricing structures and mechanisms.

We concluded that a move to a net market was unlikely to occur as:

- a majority of both fund managers and brokers indicated that, while some brokers may see advantages in selling their services net (as they may be able to maintain market share in the non-execution element of these), fund managers and brokers each had an interest in maintaining the transparency of the market, and fund managers in particular would resist this;
- such a blatant move would be implemented in the face of regulatory concern over trading opacity from both the FSA and the SEC. This could encourage regulatory resistance and further intervention in fixed income markets;

The diagram below identifies strategies that may be adopted by fund managers in response to the Proposals. These are described in Sections 8.2 to 8.7 below.



8.2 Compliance

The default response to implementation of the Proposals is for companies to comply (some reluctantly) with the new rules. This response assumes that any restructuring or procedures that are necessary to comply with the new guidance are put in place and costs associated with these are either passed onto customers, mitigated (i.e. through cost cutting measures) or absorbed.

8.2.1 Advantages and disadvantages

Advantages of compliance include the following:

- fund managers can highlight transparency gains relating to transaction costs, which may make this the preferred strategy for large UK institutional investors;
- migration of capital, labour and regulatory status is avoided; and
- the strategy is regulator friendly in the UK, and avoids incurring expenses associated with evasion, which carry the risk of regulators in other territories adopting similar proposals.

Disadvantages of compliance may be:

- the loss of automatic pass-through of costs associated with softed and bundled nonexecution services to customers, loading costs onto fund managers which may not be fully recovered through increased management fees or explicit charges;
- the potential for the cost of high quality and specialist research to rise if not sourced from overseas; and
- (if management fees rise), the potential for UK fund managers to appear more expensive than overseas rivals on the basis of fees alone, if not on a TER basis.

8.2.2 Probability analysis

Compliance is the most probable course of action for fund managers that:

- have a customer base dominated by medium or large UK-based institutional investors;
- do not have significant fund management operations overseas, increasing the cost of and limiting the practicality of adopting alternative strategies;
- believe that incremental costs can be passed on or absorbed within their cost bases; and/or
- have an operating model that does not rely heavily on softing or large quantities of broker-generated alpha research.

A high degree of compliance is likely from all segments. The greatest proportion of compliance may be amongst medium-sized UK-focused fund managers. This sub-segment may have less opportunity to circumvent the Proposals in ways that may be available to those with overseas operations or the largest UK fund managers. This segment may also be better able to absorb extra costs than smaller fund managers.

Large fund managers are also expected to choose this default option, as they have the highest number of larger UK institutional customers, which are likely to demand compliance with a regulation which, if implemented, may provide them with a net gain (see Section 7). A high rate of compliance should also occur amongst better-performing fund managers in the small segment, and UK-focused managers in all segments.

Hedge funds

The attitudes of the customers of hedge funds may differ from traditional fund managers as their customer base may be considered to be less risk-averse²³, seeking absolute returns (rather than returns relative to a market index), and thus which are more willing to pay higher management and performance fees.

Set against this, as the hedge fund industry is relatively young, many of its members have a greater reliance on (and their entry may have been encouraged by) softed services than more established fund managers. Due to their size and more recent establishment, hedge funds are also relatively mobile and thus able to migrate quickly in search of better returns.

A hedge fund manager's compliance strategy may in turn depend on the core strategy followed by the fund. Some (e.g. many long/short equity funds) rely heavily on real time idea generation and other input from brokers while others (e.g. technical strategies aimed at finding short-term stock undervaluation) have limited need for services in addition to execution.

We think it likely that a relatively high proportion of hedge funds will comply with the Proposals as a result of the ability of most to pass through incremental costs to customers. This view is also partly influenced by the current status of London as the clear European capital for hedge funds, with a 70% market share of FUM²⁴. However, to the extent that the location of a cluster in an entrepreneurial market may be volatile, wider ranges have been placed around our judgements concerning hedge funds than most other sub-segments.

8.3 Sourcing of non-execution services from other territories

A number of fund managers highlighted this option in interviews and consultation responses. It may work in one of two ways:

For larger fund managers, if a broker providing non-execution services is satisfied with the level of business it gains globally or believes more can be won, it may agree to supply services to the UK for free as part of a global commissions agreement. Thus, brokers accept a lower global commission rate, but an element of cross-

²³ The customers themselves may be identical to those of non-hedge fund managers, but choose to place a set proportion of their portfolio with a hedge fund manager to seek (more risky) absolute returns.

²⁴ Eurohedge magazine, February 2004.

subsidisation exists from the overseas customers of the fund manager to its UK customers who no longer pay for full-service brokerage.

For those unable to strike this type of deal, non-execution services could be purchased overseas. An internal recharge mechanism may then be established between the UK and overseas offices of the fund manager. The fund manager would decide whether to raise its management fee for UK-authorised business or attempt to win market share by holding it down and benefiting from any customer migration from other fund managers that do raise fees.

Smaller firms which do not have the global scope to put this strategy into operation themselves could theoretically join forces with overseas fund managers of a similar size to swap resources.

Under each of these methods of implementation, there is no movement of fund management activities or funds away from the UK.

8.3.1 Advantages and disadvantages

The most obvious advantage for fund managers able to adopt this strategy is the ability to continue to receive research and other services using the same mechanism as at present, and to avoid the potential implications of the Proposals in relation to incurring the cost of non-execution services received, negotiating the recovery of costs from customers through higher fees, and incurring any additional administrative, systems, capital or tax costs.

However, this may not represent a complete means of evasion, as some non-execution services will still need to be supplied in the UK (e.g. broker-sponsored introductions to UK companies, trade advice and one-to-one communication of ideas to UK-based staff). As a result, the re-negotiation of services received and commission levels, cost impacts, and the need to alter fees may not be avoided. Fund managers adopting the strategy may also bear the risk of

- strained relations between their UK and overseas offices (depending on form of implementation);
- incremental administrative costs of transfer of services to the UK; and
- customer resistance where benefits from the Proposals are anticipated.

8.3.2 Probability analysis

This option is predominantly available to large fund managers that negotiate receipt of services and commission levels with brokers on a global basis. Smaller companies are less likely to supply the volume of commissions to brokers to persuade them to change the way in which services are delivered, while UK-oriented fund managers may not have the appetite to evade, or (as with hedge fund managers) the overseas resources.

8.4 Partial relocation

This option involves fund managers moving enough operations overseas in order to evade the regulation. For example, dealing desks to which staff are attached that choose the brokers to place trades with and instruct on how these should be traded may be moved with minimal resource to outwardly manage funds from other territories, while strategic decisions affecting funds' performance may still be carried out in the UK The precise nature of this option, and any requirement to re-novate funds to other territories, would depend upon the final scope of the Proposals.

8.4.1 Advantages and disadvantages

If this strategy can be implemented with minimal interference to ongoing operations, its prime advantage (as with the previous strategy) is the ability to avoid incremental costs that may be incurred as a result of the Proposals and to maintain present trading relationships with brokers on a global basis.

The following may be set against potential advantages of the strategy:

- practicality and cost of implementation, including any affect that the relocation of systems and staff may have on performance;
- customer resistance where benefits from the Proposals are anticipated.
- loss of locational advantages of close working between fund managers and the dealing desk operation (this may particularly affect small companies);
- risk that other regulators may implement similar proposals, negating the strategy's benefits; and
- uncertainty as to whether the strategy will bypass FSA regulations in the long term, and whether it may lead to the adoption of a more complete withdrawal from the UK as described in section 8.5 below.

8.4.2 Probability analysis

It is likely that those fund managers with existing overseas operations (primarily European operations to facilitate the trading of European stocks) will be the main group that consider pursuing this strategy However, as the option is potentially more disruptive than that of sourcing non-execution services from overseas as described in section 8.3 above, and the main adherents of that strategy are also likely to be overseas-focused fund managers, adoption of the strategy may be limited to a small subset of this group that can implement it with the least disruption and cost.

8.5 Full relocation

Companies following this option would relocate all fund management staff from the UK and manage UK customer mandates from overseas.

8.5.1 Advantages and disadvantages

Advantages to this option are as outlined in Sections 8.3.1 and 8.4.1 above,

A key disadvantage is the expense of relocation, and the implications it would have for maintaining a UK customer base. As a result of these, it may be considered a last resort along with the options outlined in sections 8.6 and 8.7. The table below shows cost items a company wishing to relocate may need to consider and a range of illustrative costs for the relocation of 100 staff to new offices. We understand that economies of scale in relocation are of the order of 1%-2% so the costs per staff member shown may be considered to be broadly representative of relocation costs for different numbers of staff.

Table 8.1: Illustrative costs of relocating 100 staff from the UK to an overseas location						
Cost of relocating 100 staff	Min	Мах				
Penalty for terminating lease agreement Cost of preparing a new building Furniture and fittings for the new building Cost of replacing cabling/telecoms Cost of the move itself Cost of paying staff relocation allowance Cost of recruiting new staff in new location Cost of redundancy package for staff that do not want to move	£1,000,000 £1,000,000 £1,500,000 £0 £0 £300,000 £1,000,000 £1,000,000	£4,000,000 £3,000,000 £50,000 £100,000 £100,000 £700,000 £3,000,000 £2,000,000				
Total	£5,800,000	£15,850,000				
Total per staff member	£58,000	£158,500				
	Source: Deloitte analysis					

The costs shown are heavily dependent on the territory relocated to and, for companies that already have an overseas presence, may overstate costs relating to the acquisition of space, fixtures and fittings. In some territories, incentives may also be available to facilitate a move.

A further disadvantage to relocating is the risk of losing UK customer mandates as UK institutional customers in particular may prefer to retain their mandates with a fund manager complying with UK regulations. The loss of current UK customers may also limit future expansion opportunities in the UK market.

As with option 8.4 above, firms incurring the costs of moving operations out of the UK also run the risk that the country they relocate to adopts similar rules to the FSA, thus removing the benefits of the move. There may also be regulatory disadvantages to such a move with cost implications for fund managers and customers and, when moving large numbers of staff to a new location, some leakage of talent may occur.

The benefits of relocation must also be weighed against those of remaining in London. A substantial literature exists on the reasons for London's pre-eminence as a financial services centre in Europe. European Cities Monitor is an annual survey of the views of senior executives from 501 European companies on Europe's leading business cities and the key factors on which to base business location decisions. In the 2003 survey²⁵ London emerged as the top rated city in several categories seen as important to the maintenance of an international financial services cluster including:

- the availability of qualified staff;
- access to markets and international transport links;
- telecommunications; and
- number of languages spoken.

Relocating from London would forego many of the locational benefits highlighted in the report, but would also risk a loss of reputation through the lack of a presence in the longstanding hub of European financial services. The report developed the following ranking of European cities, and compared this with their position in 2002 and 1990.

Table 8.2: Ranking of E	European cities 1990, 2002 an	d 2003	
	2003	2002	1990
London	1	1	1
Paris	2	2	2
Frankfurt	3	3	3
Brussels	4	4	4
Amsterdam	5	5	5
Barcelona	6	6	11
Madrid	7	7	17
Berlin	8	9	15
Milan	9	8	9
Munich	10	11	12
Zurich	11	10	7
Dublin	12	12	-
Manchester	13	19	13
Geneva	14	15	8
Lisbon	15	17	16
		Source: Euro	pean Cities Monitor (2003)

8.5.2 Probability analysis

Given the costs and risks associated with this option, it is likely to be followed only by those that have a small UK presence and customer base or those that would otherwise be forced out of business. Given the potentially smaller aggregate impact on their cost base, if large fund managers wish to follow an evasion strategy, they are likely to source services from abroad or pursue partial relocation before fully relocating, which may be particularly expensive for this segment given numbers of staff and complexity of operations.

²⁵ European Cities Monitor, Cushman & Wakefield, Healy & Baker (2003).
8 STRATEGIES AVAILABLE TO FUND MANAGERS

Of smaller firms, high-performing companies may have little incentive to relocate, while UKfocused companies may bear a relatively high risk of customer loss and organisational upheaval. Those companies with an overseas parent and main customer base and that have found trading difficult in the UK may consider this the only option, along with those that judge they will not be able to satisfactorily re-negotiate the terms under which the package of non-execution services they currently use are supplied by brokers and third parties, and provided to customers.

8.5.3 Calculation of FUM exit

Relocation is the first strategy which may result in a material level of funds exiting the UK market. In estimating exit, we have taken account of:

- the likelihood that some companies relocating equity funds will move all other fund management from the UK, probably to a single European trading centre; and
- the magnitude of customer migration away from companies deciding to relocate towards fund managers maintaining their UK-regulated status, where benefits are anticipated from the Proposals.

To account for these opposite effects, we have assumed a 70% FUM exit from companies that decide to relocate from the UK. This is composed of the following:

- all equity funds, estimated at 54% of the total²⁶; plus
- half of the remaining assets held by relocating companies (i.e. half of 46% = 23%, 54% + 23% = 77%); less
- an assumed 10% of funds that migrate back to UK-regulated companies (10% of 77% = 7%, 77% 7% = 70%).

8.6 Sale of business

Those fund managers for whom evasion strategies above are not practical options, but may not be able to continue running a profitable business in the UK, are faced with two options: sale or exit.

8.6.1 Advantages and disadvantages

The obvious disadvantage of this strategy is that the fund manager will no longer be able to operate independently. It will therefore only be taken by those that have little other practical option, and are offered an attractive package by a prospective buyer or partner.

²⁶ Fund Management Survey, IMA (2002).

8 STRATEGIES AVAILABLE TO FUND MANAGERS

8.6.2 Probability analysis

Those fund managers that are least able to follow other options discussed above and that incur significant extra costs are expected to be concentrated among the small market segment (by number, and to a lesser extent, by value of FUM). As purchasers of their assets are likely to come from the large or medium segments, this implies some degree of consolidation in the market. In a sector where scale economies may be present, this might imply a net increase in efficiency, offset by reduced choice for consumers.

8.6.3 Calculation of FUM exit

In calculating funds that may exit the UK market, the strategy of the acquirer becomes important. We have assumed that the proportion of companies following the strategies outlined above, encompassing those where no FUM exit is forecast as well as relocation which implies significant levels of exit, is a reasonable proxy for the aggregate level of exit that might occur from the sale decisions of some fund managers. This results in an estimated 1.5% to 3.5% of funds relocating from the UK as a proportion of funds held by companies choosing this strategy.

8.7 Market exit

Under this option, fund managers that are not in a position to implement any of the strategies above exit the UK market and repay remaining assets to customers, who then need to seek other avenues to invest their money

8.7.1 Advantages and disadvantages

This option is imposed on, rather than actively chosen by fund managers. The only advantage is the minimisation of losses incurred.

8.7.2 Probability analysis

Two principle drivers may result in market exit: a lack of long-term commitment to the UK market, and/or the absence of other practical options. Fund managers with little commitment to the UK market are likely to be headquartered away from the UK while those with little option other than exit might be UK-focused but may have failed to find a buyer for their assets. Each category is likely to have found trading conditions difficult during the bear market and registered little improvement subsequently.

8.7.3 Calculation of FUM exit

While it is likely that many customers will re-invest assets with UK-regulated companies, others will choose investments in financial products other than equities, or decide not to re-invest at all. We assume below that 50% of funds returned to customers are re-invested by fund managers operating in the UK and the remaining 50% exit the market.

8 STRATEGIES AVAILABLE TO FUND MANAGERS

8.8 Conclusions on the adoption of strategies by market segment

In the table below we have input ranged probabilities of each market segment adopting each strategy. These have emerged from discussions with fund managers, brokers and other industry stakeholders, questionnaire responses and consultation responses sent to the FSA. They are also influenced by the potential cost impact on each segment as detailed in Section 6 above.

Table 8.3 - Deloitte assessment of the probability of strategy adoption by segment							
Segment /Strategy	1. Comply	2a. Source from overseas	2b. Partially relocate	2c. Relocate	3a. Sell	3b. Exit/Close	
Adoption probabilities							
	70% - 85%	15% - 25%	0% - 0%	0% - 0%	0% - 0%	0% - 0%	
Medium UK	90% - 100%	40 <i>%</i> - 50 <i>%</i> 0% - 5%	0% - 0%	0% - 0%	0% - 5%	0% - 5%	
MediumNon- UK	40% - 60%	10% - 30%	0% - 20%	5% - 15%	0% - 0%	0% - 0%	
SmallUK1	60% - 80%	0% - 10%	0% - 0%	10% - 15%	0% - 6%	0% - 3%	
SmallNon-UK1	40% - 50%	35% - 45%	0% - 0%	10% - 15%	0% - 5%	0% - 5%	
SmallUK2	40% - 70%	5% - 10%	0% - 0%	5% - 10%	5% - 20%	5% - 15%	
SmallNon-UK2 Hedge funds	30% - 40% 65% - 90%	15% - 25% 0% - 0%	0% - 0% 0% - 0%	20% - 30% 5% - 25%	5% - 10% 0% - 5%	5% - 20% 0% - 5%	
Key: Small	1 = Above avera	ge margins 200	2/03				
Small 2	2 = Below avera	ge margins 2002	2/03				
					Source: De	loitte analysis	

9. POTENTIAL IMPACT ON THE UK FUND MANAGEMENT MARKET

9.1 Approach to quantifying impact on the market

In this section, we apply the ranged probabilities of each market segment adopting the available strategies described above to estimate the effect on the UK market. This total market is estimated at £2.600bn²⁷ of FUM. In doing so, we arrive at a level of FUM that will exit the market either by relocating to serve UK customers from overseas, selling, closing or exiting the UK market completely.

9.2 Summary of impact: Funds Under Management

Table 9.1 below shows the estimated range of potential FUM exit that may occur as a result of the Proposals. In compiling the table and as discussed in Section 8 above, we have made a number of assumptions:

- Under strategies 1 (comply), 2a (source services from overseas) and 2b (partial relocation), no funds exit the UK, though under the latter strategies, some evasion of the Proposals occurs.
- Under strategy 2c (relocate), all equity FUM (54% of the total)²⁸ is initially assumed to exit the market along with half of the non-equity FUM held by those companies relocating, as some companies would withdraw from fund management in the UK altogether. However, in response to this exit, 10% of FUM is assumed to stay in the UK resulting from the migration of customers who wish to retain the UK-regulated status of their investment to fund managers that remain regulated by the FSA.
- Under strategy 3a (sale), the proportion of funds remaining in the market after accounting for Strategies 1-2c (96.5%-98.5%) are assumed to remain in the market to account on average for the strategies of the likely acquirers of companies choosing this strategy.
- Under strategy 3b (exit/close), 50% of FUM is assumed to exit the market as it is likely that a proportion of customers will respond to exit/closure by transferring to UK-regulated fund managers while others may exit or end their investments.

It should be noted that the ranged estimate of FUM exit shown below is sensitive to the above assumptions, each of which may be challenged. Our analysis is therefore arrived at

²⁷ Fund Management, IFSL (2003).

²⁸ Fund Management Survey, IMA (2002).

9 IMPACT ON THE UK FUND MANAGEMENT MARKET

on the basis of a balance of probabilities, informed by the views of fund managers, brokers and other industry stakeholders, and internal knowledge and research.

Table 9.1 – Strategy adoption probabilities and estimated exit of FUM from the UK market							
Segment /Strategy	FUM £bn	1. Comply	2a. Source services from overseas	2b. Partially relocate	2c. Relocate	3a. Sell	3b. Exit/Close
Traditional fund							
managers LargeUK LargeNon-UK Medium UK MediumNon-UK SmallUK1 SmallUK1 SmallNon-UK1 SmallNon-UK2 FUM sourced from FSA data Rest of market	612 328 307 554 57 26 58 72 2,014 538	70% - 85% 45% - 55% 90% - 100% 40% - 60% 60% - 80% 40% - 50% 30% - 40% 1,163 - 1,468	15% - 25% 40% - 50% 0% - 5% 10% - 30% 0% - 10% 35% - 45% 5% - 10% 15% - 25% 301 - 540	0% - 0% 0% - 10% 0% - 0% 0% - 20% 0% - 0% 0% - 0% 0% - 0% 0% - 0% 0 - 144	0% - 0% 0% - 0% 0% - 0% 5% - 15% 10% - 15% 10% - 15% 5% - 10% 20% - 30% 53 - 123	0% - 0% 0% - 0% 0% - 7% 0% - 0% 0% - 8% 0% - 5% 5% - 20% 5% - 10% 6 - 46	0% - 0% 0% - 0% 0% - 3% 0% - 0% 0% - 0% 0% - 5% 5% - 15% 5% - 20% 6 - 34
Total FUM exc.	0 551	1,474 —	202 604	0 100	67 166	0 50	0 10
hedge funds	2,551	1,860	302 - 004	0 - 102	07 - 150	0 - 30	0 - 42
Equity FUM exc. hedge funds	1,378	796 - 1004	206 - 369	0 - 98	36 - 84	4 - 32	4 – 23
Hedge Funds Adoption probabilities FUM allocation Equity FUM allocation	49 49 32	65% - 90% 32 - 44 21 - 29	0% - 0% 0 - 0 0 - 0	0% - 0% 0 - 0 0 - 0	5% - 25% 2 - 12 2 - 8	0% - 5% 0 - 2 0 - 2	0% - 5% 0 - 2 0 - 2
Total Market FUM	2,600	1,505 -	382 - 684	0 - 182	70 - 168	8 - 61	8 - 45
Total equity FUM	1,410	1,903 816 – 1,033	206 - 369	0 - 98	38 - 92	4 - 33	4 – 25
FUM lost from UK (£bn) Equity FUM lost Other FUM lost Total FUM lost from UK FUM lost as % of market	40.3 - 105.8 12.5 - 36.4 52.8 - 142.1 2.0% - 5.5%	0.0 - 0.0 0.0 - 0.0 0.0 - 0.0	0.0 - 0.0 0.0 - 0.0 0.0 - 0.0	0.0 - 0.0 0.0 - 0.0 0.0 - 0.0	38.0 – 92.0 10.5 – 24.9 48.6 – 117.0	0.1 – 1.5 0.1 – 1.3 0.2 – 2.7	2.2 – 12.3 1.9 – 10.2 4.1 – 22.5
Key: Small 1 =	= Above avera	ge margins 20	002/03				
Small 2 =	= Below avera	ge margins 20	002/03				
					S	ource: Delo	itte analysis

The analysis produces a range of potential FUM exit from \pounds 52.8bn to \pounds 142.1bn (2.0% to 5.5% of the market). This incorporates a central case estimate of \pounds 97.2bn of funds (3.7%) exiting.

The table below shows how this exit is distributed between market segments. It shows that of the above range that may be lost from the UK market, £24bn-£80bn is managed by the medium segment and £28bn-£62bn by the small segment (including hedge funds). The highest level of potential exit is among the small non-UK below average performance subsegment (SmallNon-UK2), where an estimated range of between 16.5% and 31.4% of FUM may exit.

Table 9.2 Estima	Table 9.2 Estimated exit of FUM by market segment							
Strategy	2a. Source from overseas	2b. Partially relocate	2c. Relocate	3a. Sell	3b. Exit/Close			
LargeUK	15% – 25%	0% - 0%	0% – 0%	0% – 0%	0% – 0%			
LargeNon-UK	40% - 50%	0% - 10%	0% – 0%	0%-0%	0%-0%			
Medium UK	0% – 5%	0% - 0%	0% – 0%	0%-7%	0% – 3%			
MediumNon-UK	10% – 30%	0% - 20%	5% – 15%	0% – 0%	0% – 0%			
SmallUK1	0% – 10%	0% - 0%	10% – 15%	0% - 8%	0% – 0%			
SmallNon-UK1	35% - 45%	0% - 0%	10% – 15%	0% – 5%	0% - 5%			
SmallUK2	5% - 10%	0% - 0%	5% - 10%	5% — 20%	5% – 15%			
Smallinon-UK2	15% - 25%	0% - 0%	20% - 30%	5% - 10% 0% 5%	5% - 20%			
	0 /0 - 0 /0	076 - 076	570 - 2570	0 /0 - 5 /0	0 /0 = 3 /0			
Potential FUM	FUM exiting –	FUM exiting –	FUM exiting	Total FUM	% of FUM			
exit (£DII)	strategy 20	strategy sa	strategy 30	exiting	exiting			
LargeUK	0.0 - 0.0	0.0 - 0.0	0.0 - 0.0	0.0 - 0.0	0.0% - 0.0%			
LargeNon-UK	0.0 - 0.0	0.0 - 0.0	0.0 - 0.0	0.0 - 0.0	0.0% - 0.0%			
Medium UK	0.0 – 0.0	0.0 – 1.2	0.0 - 5.8	0.0 – 7.1	0.0% – 1.8%			
MediumNon-UK	24.4 – 73.4	0.0 - 0.0	0.0 - 0.0	24.4 – 73.4	3.5% – 10.5%			
SmallUK1	5.0 – 7.5	0.0 – 0.3	0.0 - 0.0	5.0 – 7.8	6.9% – 10.8%			
SmallNon-UK1	2.3 – 3.5	0.0 – 0.1	0.0 – 0.8	2.3 - 4.4	6.9% – 13.2%			
SmallUK2	2.5 – 5.1	0.1 – 0.7	1.8 – 5.5	4.4 – 11.3	6.1% – 15.4%			
SmallNon-UK2	12.6 – 19.0	0.1 – 0.4	2.3 – 9.1	15.0 – 28.4	16.5% – 31.4%			
Hedge Funds	1.7 – 8.5	0.0 - 0.1	0.0 – 1.2	1.7 – 9.8	3.5% – 20.1%			
(£000)	48.56 – 117.0	0.2 – 2.7	4.1 – 22.5	52.8 – 142.1	2.0% – 5.5%			
Key: Small 1	= Above average m	argins 2002/03						
Small 2	= Below average m	argins 2002/03						
	-			Source	: Deloitte analysis			

It should be noted that, as some of the FUM exit is accounted for by unprofitable (and potentially inefficient) firms, their exit may be considered a net benefit to the industry / economy, as it might allow for the re-direction of resources to more profitable activities. Offsetting this however, our analysis has been conducted at a particular point in time in a depressed market, such that these companies may return to or move into profitability as market conditions improve.

10. IMPACT ON THE UK ECONOMY

10.1 Introduction

In this section, we extend the analysis performed to estimate the impact on the fund management market to assess at a high level the potential impact of the Proposals on the UK Economy. This analysis is based on the estimated range of the percentage of total FUM leaving the UK following implementation of the Proposals.

10.2 Methodology

In order to estimate the wider impact of the Proposals, we first need to assess the total contribution of the fund management industry to the UK economy.

The contribution to the economy of each individual producer, industry or sector is measured using Gross Value Added (GVA), which, in this case, is calculated as the sum of the industry's profits and its total employment costs. The GVA of the fund management industry is indicatively estimated at Section 10.3 and the GVA loss resulting from the exit of funds estimated at Section 9 is shown at Section 10.4.

A change in the economic activity of an industry, such as the relocation or exit of a proportion of funds from the country, will have a direct impact on the economy and an indirect impact via its affect on closely related industries and other parts of the supply chain. Direct and indirect impacts are measured using economic multipliers.²⁹ We use an employment multiplier to estimate the wider economic impact in terms of GVA loss of an exit of funds and relate this to total economy GVA, and by extension to total economy Gross Domestic Product (GDP) at Section 10.5.

10.3 Calculation of industry-specific GVA

GVA may be calculated by adding the economic profit of an industry (as distributed among owners and shareholders) to employment costs (as distributed among employees). However, as noted at Section 6.8 above, financial data sourced from the FSA provides margins resulting from a level of earnings including depreciation. This has been used as a proxy for, and may overstate, measures of economic profit usually associated with GVA and is a key reason why our estimates of the value of industry GVA, and that of GVA lost, should be taken as indicative. Using the FSA database to arrive at an average margin, total profits in 2002/03 may be estimated at £1.82bn on an industry cost base of £11.48bn.

²⁹ Induced effects resulting from changes in household expenditure may also be estimated using multipliers, however the accuracy of these may be considered to be less reliable and they are not published by the Office of National Statistics.

10 IMPACT ON THE UK ECONOMY

International Financial Services London (IFSL)³⁰ has estimated that employment costs are around 55% of total costs in the institutional market and 60% in the retail market. Accepting an IMA estimate³¹ that the institutional market is nearly 15 times larger than the retail market, total institutional costs may be estimated at £10.76bn with employment costs accounting for £5.92bn, while retail costs can be estimated at £0.71bn with employment costs of £0.43bn. This gives a total employment cost for the industry of £6.35bn.

The GVA of the fund management industry, as the sum of total profits and employment costs, was therefore around £8.17bn in 2002/03, accounting for approximately 16.3% of the financial services industry and 0.9% of total economy GVA.

Table 10.1 - Calculation of industry-specific GVA	
Total costs (£bn)	£bn 11.5 ⁷
Institutional funds Retail funds	10.8 0.7
From which we obtain:	
Total employment costs	6.3 ²
Institutional funds Retail funds	5.9 0.4
Total profits	1.8
Fund management industry GVA (employment costs + profits)	8.1 ³
as % of financial services GVA as % of the whole economy GVA	16.3% ⁴ 0.9% ⁵
¹ The institutional market is assumed to be 15 times larger than the retail market ² Employment costs are 55% of total costs in the institutional market and 60% in the retail market ³ Total industry-specific GVA equals total industry profits + total industry employment costs ⁴ 2002/03 financial services GVA was £50.1bn ⁵ 2002/03 whole economy GVA was £909.8bn Sources: ESA Database JESL JMA ONS and Deloi	tta analysis
Sources. FSA Database, IFSL, IMA, ONS and Deloi	tte analysis

10.4 Estimation of GVA lost

As described in section 9.2 above, we estimate that, if the Proposals are implemented as drafted, a range of FUM from 2.0% to 5.5% may exit the UK market. Estimates presented in the table below show that this would reduce GVA attributable to fund management by between 1.9% and 4.8%, partly due to a lower level of profitability amongst those that exit when compared with the industry average, as indicated by the segmental analysis of exit at Table 9.2 in the previous section.

³⁰ Fund Management, IFSL (2003).

³¹ Fund Management Survey, IMA (2002).

10 IMPACT ON THE UK ECONOMY

Table 10.2 - GVA lost as a result of FUM exiting the market						
	Low end (£m)	Central case (£m)	High end (£m)			
Fund management GVA (= Total profits + Employment costs)		8,143				
Estimated profits lost	24	39	44			
Estimated employment costs lost	128	236	346			
Estimated GVA lost	152	275	390			
GVA lost as a proportion of fund management GVA	1.9%	3.4%	4.8%			
		Source: ONS,	Deloitte analysis			

10.5 Estimation of wider economic loss using an employment multiplier

As noted above, direct and indirect impacts on the economy stemming from a change in activity in a particular sector may be measured using economic multipliers. There are three types of multipliers:

- output multipliers, measuring the effects of a change in demand;
- employment multipliers, measuring the effects of changes to employment; and
- income multipliers, measuring the effects of changes to income.

Table 10.2 above shows that the main result from an exit of funds from the UK will be a change to levels of employment in the industry. Though changes to output and income may occur, these are impossible to predict with any certainty. The Office of National Statistics (ONS) produces employment multipliers for each sector of the UK economy. Though a specific multiplier is not produced for the fund management industry, a multiplier can be estimated by weighting available financial services sector multipliers by their contribution to the sector's GVA. This is shown in Table 10.3 below.

Table 10.3 - Estimation of the employment multiplier						
Industry	GVA (2003 prices)	% of total GVA	Employment multiplier			
Banking and finance	£30,309,546,564	61%	1.86			
Insurance and pension funds	£11,491,311,084	23%	2.92			
Auxiliary financial services	£8,278,164,501	17%	1.86			
FM industry employment multiplier (weighted average):			2.10			
		Source: ONS, I	Deloitte analysis			

Application of this multiplier effect to the GVA loss shown in Table 10.2 results in a GVA loss to the wider economy of between £321m and £819m as shown in the table below.

10 IMPACT ON THE UK ECONOMY

Table 10.4 – Estimated lost GVA following application of an employment multiplier						
	Low end (£m)	Central case (£m)	High end (£m)			
Estimated GVA lost	152	275	390			
Employment multiplier	2.1	2.1	2.1			
Estimated GVA lost accounting for wider economic impacts	321	578	819			
		Source: ONS,	Deloitte analysis			

This level of GVA loss represents between 0.035% and 0.090% of UK economy GVA of £910bn in 2002/03. GDP is obtained from GVA by adding taxes paid to and subtracting subsidies received from the Government. As industry-level economic activity is not normally expressed in terms of GDP, the ratio of industry GVA lost to whole-economy GVA is more meaningful than comparing industry GVA with whole-economy GDP. However, this ratio has also been noted in the table below for completeness.

Table 10.5 – Estimated exit as a proportion of 2002/2003 GDP						
	Low case (£m)	Central case (£m)	High case (£m)			
GVA exiting the economy	321	578	819			
2002/03 GVA		909,827				
Exit as proportion of 2002/03 GVA	0.035%	0.064%	0.090%			
2002/03 GDP		1,054,061				
Exit as proportion of 2002/03 GDP	0.030%	0.055%	0.078%			
		Source: ONS	Source: ONS, Deloitte analysis			

SUMMARY OF METHODS USED TO ESTIMATE THE SIZE OF THE MARKET FOR SOFTED AND BUNDLED SERVICES

A1 Introduction

This section summarises the three methods that were used to arrive at an estimate of the size of the market for softed and bundled services in 2002/03. These are "top-down" approaches which first estimate the total level of commissions paid by UK fund managers to brokers for UK and overseas equity trades. Total commissions can be multiplied by the proportion of commissions attributable to softed and bundled services (as calculated at Section 3.2 of the main report) to arrive at a value of commissions paid for softed and bundled services.

A1.2 Method 1: Value of equity trades

Introduction

This approach requires the estimation of the total value of non-proprietary equity trades (i.e. those placed by brokers for third parties such as fund managers) passing through both the London Stock Exchange (LSE) and exchanges in other geographic markets. The average level of UK commissions and a weighted average level of commissions paid on overseas markets can then be applied to this value to gain the value of total commissions paid by UK fund managers in the UK and overseas.

Data used

Data was obtained from the LSE showing that non-proprietary trades (i.e. those trades placed by brokers for third parties) in UK equities placed on the exchange totalled £1,112bn in 2003³². This total was reduced by 48.7% to account for the proportions of LSE trades carried out by brokers on behalf of overseas clients and on behalf of UK clients placing trades without the assistance of a fund manager (both private clients and companies)³³. The exclusion of these trades results in an estimated value of non-proprietary trades in UK equities undertaken for UK fund managers of £570bn.

Elkins McSherry data was obtained showing an average UK commission rate for all equity trades in 2003 of 13.87bps. This was applied to the value of trades above to arrive at an estimate of 2003 commissions paid by UK fund managers for trades in UK equities of £0.79bn.

³² Secondary Market Factsheet, London Stock Exchange, (December 2003).

³³ Survey of London Stock Exchange Transactions 2000, London Stock Exchange (2000).

To obtain a value for commissions paid by fund managers for trades in overseas equities placed on other exchanges, we used IMA data³⁴ reproduced below showing fund managers' allocation of equities by world trading region.

Table A1.1 – Fund managers' allocation of assets between equity markets					
2002					
	Allocation (%)				
	. ,				
UK	36%				
Europe	17%				
North America	31%				
Japan	9%				
Emerging Markets	5%				
Other	2%				
Total	100%				
	Source: IMA				

The table shows that, on average, 36% of fund managers' equity portfolios were held in UK equities, a large majority of which are traded on the LSE. The £570bn value of trades placed in UK equities on the LSE may therefore be assumed to represent around 36% of total trades placed by fund managers, with trades in overseas equities placed on other exchanges accounting for the remaining 64%. This gives an approximate value of trades placed on other exchanges of £1,014bn.

As with UK trades above, Elkins McSherry data was used to source average commission levels in world equity markets. The table below shows average commissions for the regions shown in Table A1.1.

Table A1.2 – Average commissions in	world equity markets 2003 Average commission (bps)
UK Europe North America	13.87 18.73 21.32 13.05
Emerging Markets Other	34.78 3312 Source: Elkins McSherry ³⁵

When the average commissions shown at Table A1.2 are weighted by the proportion of equities held in each region shown in Table A1.1, a weighted average overseas commission rate of 20.58bps results. When this is multiplied by the £1.01bn value of overseas trades, total commissions paid for trades in overseas equities of £2.09bn result. When added to UK commissions calculated above of £0.79bn, total commissions paid by UK fund managers to brokers may be estimated at approximately £2.88bn in 2003.

³⁴ Fund Management Survey, IMA (2002). The IMA sample included responses from 55 member firms, accounting for 90% of assets managed in the UK.

³⁵ The average commissions shown have been aggregated from country data supplied by Elkins McSherry. UK average commissions represent an average between buys and sells. We understand from Elkins McSherry that zero commission trades are included in the average but have no material effect upon it.

A1.3 Method 2: Imputed from stamp duty

Introduction

Fund managers must pay stamp duty of 50bps on the purchase of shares. As commission is charged on both purchases and sales of shares, if it is assumed that fund managers buy and sell equal amounts of shares, it may be further assumed that stamp duty receipts should have an approximate relationship to the total commissions paid to brokers for trades placed in the UK equal to:

Total commissions from UK trades = Stamp duty receipts x (Average commission rate x 2 / stamp duty rate).

Data used

Stamp duty receipts relating to equity trades settled through CREST³⁶ in financial year 2002/3 totalled £2.25bn.³⁷ Using the formula above and Elkins McSherry data on average UK commissions in 2003, commissions from UK trades may be estimated at £1.25bn as shown below:

£2.25bn³⁸ x (2 x 13.87bp / 50bp) = £1.25bn

If total commissions earned outside the UK are calculated as in method 1 above, totalling $\pounds 2.09$ bn, total UK and overseas commissions may be estimated at $\pounds 3.33$ bn in 2002/03.

A1.4 Method 3: Fund activity analysis

Introduction

Total commissions may also be estimated by taking the total equity funds under management in the UK and examining the trading turnover of these equities over a year in the world equity markets in which they are held to give a total value of equity trades. Average commissions data may then be used to arrive at total commissions paid on this value of trades.

³⁶ CREST provides settlement services for the London Stock Exchange, Irish Stock Exchange, virt-x and a range of Electronic Crossing Networks

³⁷ Source: Inland Revenue.

³⁸ We understand that intermediaries (e.g. brokers) generally benefit from relief for stamp duty and so intermediary trades should not inflate stamp duty receipts for the purposes of this approach. We further understand that trades placed by overseas institutions may be included in the value of receipts, but that these should be offset to some extent by stamp duty relief enjoyed by fund managers on Contracts for Differences (CFDs) including underlying equity trades. Data was not available to separate out these components of stamp duty take from total receipts.

Data used

Total funds under management in the UK at the end of 2002 were £2,600bn³⁹. Around 54% of UK funds are composed of equities⁴⁰ giving total equity funds under management of £1,404bn. These are allocated between equity markets as at Table A1.1 above. The turnover of equities placed in each market was obtained from statistics purchased from WM Company showing average activity levels of UK pension funds in each geographic region. Finally, average commissions in each region were again taken from Elkins McSherry data (see Table A1.2) to arrive at a total level of commissions paid for equity trades in each region. These are then added to arrive at a total level of commissions paid of £3.44bn in 2003. This approach is summarised in Table A1.3 below.

Table A1.3 Estimated commissions spend in world equity markets							
bps	Allocation (%)	Funds under management (£)	Activity level* (%)	Average commission (bp)	Total commission (bp)		
UK 36% 505 98% 13.87 0.69 Europe 17% 239 123% 18.73 0.55 North America 31% 435 160% 21.32 1.48 Japan 9% 126 146% 13.05 0.24 Emerging markets 5% 70 153% 34.78 0.37 Other 2% 28 153% 23.12 0.10 Total 100% 1,404 3.44 3.44 * Activity represents the element of turnover in excess of the net investment during the period (i.e. voluntary dealing). Formula = Purchases + Sales - Net Investment / Average Capital Employed.							
Net Investment denotes the Modulus of the Net Investment (e.g. If Net Investment = -6,000 then Net Investment = 6000, If Net Investment = 6,000 then Net Investment also = 6,000.							
Average capital employed is defined as the initial market value plus time-weighted investment (i.e. if the initial market value is 1,000 and there has been a net investment of 200 on day 3 of the month (taking a 31 day month) then the average capital employed = 1000 + ([31-3]/31 *200) = 1180.65.							
			Source: Deloit	e fund managers	questionnaire		

³⁹ Fund Management, IFSL (2003).

 $^{^{40}}$ Fund Management Survey, IMA (2002).

FUND MANAGERS' VIEWS ON THE FUTURE OF THE RESEARCH SECTOR

A2 Introduction

This section summarises the answers of respondents to our questionnaire focusing on questions aimed at forecasting the potential environment for the provision of research if the Proposals are implemented. We separated research, which may be received in written or verbal form, into the following categories:

Proprietary / non-proprietary - proprietary research is broadly defined as research that is requested by the fund manager on equities of their choice. This may be following advice from a broker, but the decision to source the research is taken by the fund manager. In contrast, non-proprietary research is pushed by the provider to the fund manager.

Type of provider - broker / independent research company or boutique.

Type of equity - main stocks / niche and small cap stocks

A2.1 Volume and value of research received

At present, research provided by brokers accounts for around 52% of all research sourced by volume, of which non-proprietary accounts for 33% and proprietary 19%. In-house research accounts for about 45% of research provision, while the remaining 3% is provided by independent research companies and boutiques.

In-house and proprietary research are highly valued due to their greater focus, while independent research is valued higher than material from brokers, particularly research on niche or small cap companies. Some fund managers indicated that they would buy more independent research if the supply was present in the market. Table A2.1 summarises the findings of the questionnaire on volume and value of research purchased.

Table A2.1 – Volume and value of research currently produced										
	In- house	P Broker /main stocks	Proprietary Broker niche/ small- cap	y Ind. main stocks	Ind. niche/ small- cap	Broker /main stocks	Nor Broker niche/ small- cap	n -propriet Ind. main stocks	a ry Ind. niche/ small- cap	Total
Volume produced / sourced (as % of total)	45.2%	18.1%	0.4%	1.4%	0.2%	29.4%	3.7%	0.7%	0.8%	100%
Value (out of 100)	85	75	75	82	85	29	46	30	70	
	Source: Deloitte fund managers questionnaire									

We also asked fund managers how their take-up of research had altered over the past ten years and how they would envisage it changing after implementation of CP176. Analysis of

responses to these questions is based on a smaller sample than that used for the table above as fewer firms felt confident in forecasting the future research environment, and fewer were able to provide historic data, than could indicate their current requirements.

Respondents' views on the volume and value of research provided 10 years ago are shown in the table below.

Table A2.2 – Vo	olume and	l value of	research	produce	d 10 year	s ago				
		F	Proprietar	у			Nor	n-propriet	tary	
	In- house	Broker /main stocks	Broker niche/ small- cap	Ind. main stocks	Ind. niche/ small- cap	Broker /main stocks	Broker niche/ small- cap	Ind. main stocks	Ind. niche/ small- cap	Total
Volume produced / sourced (as % of total)	31.6%	9.9%	1.0%	0.4%	0.1%	49.6%	5.6%	0.6%	1.1%	100%
Value (out of 100)	77	56	25	35	35	40	50	0	62	
					S	ource: De	loitte fun	d manage	ers questi	ionnaire

Over the past ten years, non-proprietary broker main stock research has fallen from 50% of volume produced to 29%, while production of in-house research has increased by 13 percentage points from 32% to 45%. Independent research has risen from 2.2% to 3.1% of the market, coinciding with an increase in its value to fund managers (and that of proprietary research in general) indicating that fund managers may now have more control over the type of research received.

Expected volume and value of research produced if the Proposals are implemented is shown in the table below.

Table A2.3 – Vo	lume and	I value of	research	produce	d if the Pi	oposals a	are imple	mented		
		P	roprietar	у			Nor	n-propriet	ary	
	In- house	Broker /main stocks	Broker niche/ small- cap	Ind. main stocks	Ind. niche/ small- cap	Broker /main stocks	Broker niche/ small- cap	Ind. main stocks	Ind. niche/ small- cap	Total
Volume produced / sourced (as % of total)	64.6%	8.4%	1.0%	0.9%	0.8%	20.1%	2.9%	0.2%	1.1%	100%
Value (out of 100)	94	80	72	32	82	40	50	0	62	
					S	ource: De	loitte fun	d manage	ers questi	ionnaire

Fund managers expect that implementation of the Proposals would result in a further shift towards in-house production, rising from 45% to around 65% of total research sourced, and that this would be valued more than at present. This rise is offset by a fall in market share attributable to broker-produced research from 52% to 32%, while independently produced research is expected to maintain its 3% share. A small rise in the share of niche and small-cap research is expected from 5.2% to 5.7%.

The results shown in Table A2.3 may not indicate that larger fund managers expect to produce much higher quantities of research in-house but rather that a higher proportion of a smaller total may be sourced in-house as the level of broker research produced decreases. Feedback from fund managers with limited or non-existing in-house capabilities suggests that they will adopt a wait-and-see strategy in order to understand changes brought about by the Proposals before committing to large investments in in-house provision.

A2.2 Potential free rider problems

When assessing the impact of unbundling on the demand for research, Charles River Associates⁴¹ predicted a "significant free rider problem" resulting from the "inability of fund managers to effectively discriminate between those customers that are willing to pay for research and those that are not." It concluded that this combined with "an unwillingness for any increase in annual management charges to reflect the cost of advice" will mean that "the equilibrium level of research will fall below the optimum."

We asked fund managers to what extent they agree with the above argument. Out of 14 respondents that answered, 75% strongly agreed or agreed while 25% neither agreed nor disagreed. This level of acceptance of the argument was accompanied by comments such as "it would be extremely difficult to apply the knowledge and information gained from research or bundled services only to customers that have paid for these services when making investment decisions for the whole fund."

A2.3 Spending on research

In its CP176 cost-benefit analysis, OXERA commented that "total spending on research may fall, but spending on in-house research, and the type of third party research currently paid for with hard money, is likely to increase proportionally, if not absolutely. As a result, general competitive conditions are likely to improve, which should be advantageous to the customers of research services."

We asked respondents what their level of agreement with the above statement is. 23% of respondents strongly disagreed, and a further 38% disagreed, while 31% neither agreed nor disagreed, and 8% strongly agreed.

Comments received suggest that fund managers doubt that smaller companies would be able to either set up in-house research capabilities or buy sufficient quantities of unbundled research from brokers and independent providers.

^{41 &}quot;An assessment of the proposed changes to regulation of bundled brokerage and soft commission arrangements", Charles River Associates Ltd, (October 2003).

A2.4 The future of the independent research sector

We asked respondents if they agree with OXERA's argument that "when research is purchased with hard cash rather than bundled, the independent research sector will be able to flourish as it will compete on a level playing field and, in particular, be able to provide better quality research in small cap and niche stocks."

In total, 27% of respondents agreed with the statement, 20% neither agreed nor disagreed, and 53% disagreed or strongly disagreed. Those disagreeing suggested that, depending on the form of implementation of Proposal 1, independent researchers could lose access to soft credits but, as fund managers may prefer not to use hard money to source bespoke research, commission sharing agreements might become the only source of income for the independent sector.

THE FUND MANAGERS QUESTIONNAIRE

This section contains the questionnaire that was sent to fund managers.

npact of FSA Consultation Paper Chin Mone CP176 on Incremental costs and m02070077890.ext (increased) Competitiveness competitiveness	Fund Managers Questionnaire	Calm Themas Parter Parter Parter Delorts Seconter Court 1 Seconter Court 1 Seconter Sizer London FCA.4TK Oreston relating to the content of the questionnaire can be addressed to:	This questionate has been complete by Delsine to solice the optimum of Fund Managers rel- tes the impact on their costs and competitiveness of the proposals contained within Consultation Paper CP136. Responses to the questionnaire should be sent so:	the induction between on their cases and competitioneess of the perpenditure transformation between the transformation between tr	Deloitte. Fund Managers Questionnaire pact of FSA Consultation Paper CP176 on Incremental costs and competitiveness January 2004
	apact of FSA Consultation Paper Chin Mone and another and another and another and another and competitiveness	Fund Managers Questionnaire npact of FSA Consultation Paper CP176 on Incremental costs and competitiveness	Delotte Contours Delotte Contours Pund Managers Questionnaire Contraction Indet of FSA Consultation Paper Contraction Operation Contraction Operation Contraction Indet of FSA Consultation Paper Contraction Competitiveness Contraction		January 2004

INTRODUCTION

The FSA has asked us to examine the impact of CP176 on the UK Fund Management industry. They have specifically requested that we assess incremental costs that may arise us a result of the regulation and strategies that may be followed by the industry to mitigate extra costs. They are also interested in the extent to which any extra costs can be passed through to trustees and other clients.

GENERAL OUESTIONS

Throughout this survey, two types of commission arrangements are referred to:

- Soft commission arrangements are those in which a fund manager, by agreeing to send trades to a broker, receives, in addition to 'pure' trade execution, credits ('soft deflues') which can then be used to purchase services such as research and information services. .
- Bundled brokerage arrangements are those in which a fund manager, by agreeing to send trades to a broker, receives, in addition to 'pure' trade execution, other services, chiefly research, from the broker. .

Please note that the questions below are not in strict numerical order as they have been selected from a larger pool. Those questions excluded from the questionnaire may be discussed with you separately at a later date.

Part A: Background information Name of firm and business unit

Participants at meeting

Where questions below do not state that data from a specific year should be used in your answer, please include data from the latest available year. Please give the year you will use below and state whether this is a calendar year or a financial year. Position Name

calendar year? financial year (please erase) Vest Please give all financial information required in this questionnaire in pounds sterling.

e:

Part B: General questions

B.1 Please complete the following table showing the trend in commission paid to brokers over the last five years

sdq	2002	2001	2000	1990	1998
Commission received on standard trade					
Commission received on programmed trades					
Commission received on difficult trade (e.g. including working the trade)					
Average commission received					

B.2 Can you provide an explanation of the trend shown and the extent to which specific events caused step-changes in commission paid?

B.4 Please complete the tables below showing trades with, and commission puid to, UK and overseas brokers in the past three years.

	2002		
(£)	UK regulated brokers	Overseas regulated brokers	TOTAL
Trades by value			
Commission paid			
mmission paid where softed credits received			
alue of soft credits received			
Commission paid where bundled services received			
Estimated value of bundled services received			

Other impacts:	C. Definition of Definition of the second se		B. 7 If CP176 is implemented, do you anticipate receiving:	 bundled execution and non-execution services, rehating the non-execution element of these to continues. 	 b) separated execution and non-execution services, where non-execution services are purchase for a surgle bundled price; 	 a menu of separately proved execution and non-execution services (e.g. the tautil proving model); a missione of the above depending on broker / efferte? 	 where other artingements (please specify). 	Please state attricitated percentage of each and state if this would depend on the client (e.g.	Remaining (Remaining (Remaining) Remaining (Remaining)	Please coplain your answer:			It & Who would drive this decision? (e.g. broker, fund manuger, client)	e h B.9 Wondd you anticipate explicitly charging for non-concution services when you invoice clients	or making an adjustment to your management feel? Please state if this would depend on the client (e.g. mathational / netail)		B. 10 Who would drive this decision? (e.g. fund manager, client).	
														and above sheet with	viour? For			
Γ	TOTAL						Γ	TOTAL			1	-		on of CP176 over attach separate	your trading behav			
	Overseas regulated brokers. TOTAL							Overseas regulated brokers TOTAL						proposed implementation of CP176 over consultation? (please attach separate	eve this would change your trading behav	see/failt' le for this risefailt' of stocks?		
2001	UK regulated Overseas TOTAL brokers regulated brokers		ed by	2			2000	UK regulated Overseas brokers regulated brokers TOTAL		ed	1			ats regarding the proposed implementation of CP176 over onse to the FSA consultation? (please attach separate	, how do you believe this would change your trading beha	rades carried out rise / full? ould be responsible for this rise/full? to different types of stocks?		

က
\mathbf{X}
ð
Ĕ
Б
∢

B.11 To what extent would you agree with the cost of components of a hundled / softed trade booked with UK / overseas brokers (as published by OXERA in its analysis for the FSA on the costs and henefits that might accure from CP176/?

faint and a lot of the Realsdown of

	OXERA analysis	UK brokers	Overseas hrokers
Average commission rate	145p		
Execution only (electronic trading platform)	5-Sbp		
Working a trade	2-5bp		B-2 - 140
Soft commission	Ibp		-
Bundled services	3bp		

Please explain your answer

B.12 If the proposals contained in CP176 were implemented such that softed services currently received were further restricted and the price of non-execution services was unbundled from execution services, please indicate in the table below the extent to which you believe a full in broker commissions will offset the value of softed credits and bundled services.

Total value of softed services received	From Table C.4
Estimated fall in broker commission to compensate (£ or %)	
Estimated value of blandled services received	From Tables C.4 and C.9
- Estimated full in broker commission to compensate (E or %)	

Please explain your answer, where possible giving examples from previous price negotiations.

B.14 Could you provide examples and supporting evidence of when brokers were able to impose unilateral changes to the terms of their contracts with fund managers and/or increase their charges / not pass through costs in the past? Why were fund managers not able to constrain these changes?

B.16 With respect to current developments in the Fund Managers market (e.g. step out arrangements), to what extent have you taken these up and what influence will they have on the proposals contained in CP176 if implemented?

1. Questions relating to Proposal 1 in CP176

La Please complete the table below to show for UK-reglated business :

- the value of services received through or paid for using soft credits (L); and the percentage of services received by value that were purchased or provided by the issuer of the soft credits (e.g. if 50% were provided by the broker and a further 20% purchased by the broker on your behalf, 70% of services were purchased or provided by the issuer of the credits). . .

Please note:

- The value of research received through or puid for using soft credits is covered later at question R.f and should not be included here. If insufficient information is available to disaggregate the services received d paid for using .
- soft credits, please complete the "total" rows only. .

	2002	2001	2000
Market Prichtg and Information services (e.g. terminals)			
The purchased or provided by issues on your behalf			
Hardware software & other equipment			
"+ purchased or provided by issuer on your behalt			
Valuation and performance measurement services			
To purchased or provided by must on your behalf			
Other services			
The purchased or provided by issuer in your behalf			
Total value of services received			
I stal % purchased or provided by issuer on your behalf			

က
\mathbf{X}
ð
ũ
Ч
∢

1.b What is the percentage of brokers you trade with that provided you with these services in the last financial year? [1.c. number of brokers providing you with these services divided by the total number of brokers you trade with]

1.1 MPIS (Market Prising Information Systems, e.g. Bloomberg)

1.1.1 In the last year, which MPIS services do you receive through softing arrangements?

1.1.4 Has your take-up of these services (through both sofied and non-sofied arrangements) altered over the past 3 years?

1.1.5 If CP176 is implemented, please indicate:

- .
- .
- How the quality of these services may change: (taking into account any changes in quality) how the average price of these services to you may change (e.g. new price value of soft credits previously used); whether you will be able to pass the cost of these services through to your clients, and (taking into account the above) the quantity you would expect to purchase relative to the
 - quantity currently obtained.

	Service I	Service 2	Service 3	Service 4	Service 5
Average quality	up/down/same	up / down home	up / down (same	up/down/same	up / down (same
Average price	(*a change)	(%- change)	(% change)	(**=change)	(%cchange)
(ed. 1040)					
Cost recovery	(*+ change)	(% change)	(% change)	(%+ change)	(%, change)
(c.g. 100% recovery)					
Amount purchased	(% change)	(% change)	(% change)	(% change)	(%, change)
(5,01+,722)					

Please detail reasons and (to the extent possible) supply evidence supporting your answer.

L2 Hardware,

1.2.1 In the last year, which hardware, software & other equipment do you receive through softing arrangement?

1.2.4 Has your take-up of these services (through both softed and non-softed arrangements) altered over the past 3 years?

1.2.5 If CP176 is implemented, please indicate:

- •
- .
- How the quality of these services may change; (taking into account any changes in quality) how the average price of these services to you may change (e.g. new price value of soft treedis previously used); whether you will be able to pass the cost of these services through to your clients, and (taking into account the above) the quantity you would expect to purchase relative to the
 - quantity currently obtained.

Average quality up (down sam Average price (% change) (c.g10%)	are wp/down/same	were ficknessed in service	The second second second second	
Average price (% charge) (e.g(0%)		141 0111011-001100	anne anno da	up down sume
Average price (% charge) (e.g10%) Control (% charge)		5		
(e.g10%) (1	(** change)	(** change)	(%+ change)	(%change)
Post measure (%) channel		1		
CONCIDENTS	(%s change)	(*4-change)	(*+ change)	("is change!
(e.g. 100% recovery)		Sector Sector		
Amount purchased ("-, change)	(% change)	(%- change)	1%+ change!	(States of the second s
15601+ 2.4)				

Please detail reasons and (to the extent possible) supply evidence supporting your answer.

ო	
×	
Z	
T	
Å	

	- 6
	-
	2
	7
	-3
	5
	-
	-5
	5
	2
	1
	2
	2
	-5
	s
	4
	4
	5
	5
	5
	÷
	2
	£
	2
	2
- 1	
	τ
	7
	3
	1
	_
	1
	1
	2
	e.
	3
	2
	12
	2
	÷
	12
	8
	-
	17
	_

1.3.1 In the last year, which valuation and performance measurement services do you receive through softing arrangements?

1.3.4 Has your take-up of these services (through both softed and non-softed arrangements) altered over the past 3 years?

1.3.5 If CP176 is implemented, please indicate:

- How the quality of these services may change: (taking into account any changes in quality) how the average price of these services to you may change (e.g. new price value of soft credits previously used). whether you will be able to pass the cost of these services through to your chents, and (taking into account he above) the quantity you would expect to purchase relative to the quantity currently obtained.

	Service 1	Service 2	Service 3	Service 4	Service 5
Average quality	up / down (same	up - down sume	up / down (same	up/down/same	up / donn / same
The second second	A DESCRIPTION OF A DESC		The second second	The second s	A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR A
Average price	(*+ change)	("% change)	(*+ chunge)	(%+ change)	(** chinge)
(#g10%)					
Cost recovery	(%+ change)	1% change!	("= change)	("+ change)	(%change)
(e.g. 100% necovery)					
Anount purchased	(%) change)	(%change)	(%) change)	(%" change)	(% change)
(# St +10%)					

Please detail reasons and (to the extent possible) supply evidence supporting your answer.

2. Questions relating to Proposal 2 in CP176

2.a On average, what percentage of total commission paid corresponds to the following elements of a typical trade?

Element of trade	Percentage of commission
Trade (execution-only)*	6
Softed services	.45
Bundled services (e.g. working trade).	
TOTAL COMMISSION	1002
100 100 100 100 10 10 10 10 10 10 10 10	Chicken strategies in the many stations

* e.g. straightforward trade that could be executed by execution-only broker

2.b With reference to your answer above estimating the proportion of broker commission currently accounted for by bundled services, on average, to what percentage of the bundle you receive in a typical trade do the following bundled services correspond? (Please complete the following tuble)

Will incur a charge if unbandled?	Yes / No	Yes/No	Yes / No	Yes/No	Yes / No	Yes/ No	Yes / No	Yes / No	Yes/No	Yes / No Rescription	$\gamma_{\text{des}} \wedge No$	Yes / No	
Is it inseparable from trade?	$\gamma es/N\sigma$	Yes/No	Yes/No	Yek) No	Yes / No	Yes / No	$\delta N/S^2 \Lambda$	Yes/No	Ves / No	Yes/No	Yes/No	Yes/No	
% of hundle received	*	2	4	4	1	×.	4	2	5	8	2	14	100%
	Working a trade	Trade advice	Broker advice research: proprietary MAIN STOCKS	Broker advice/research proprietary NICHE / SMALL-CAP	Independent/boutique proprietary. MAIN STOCKS	Independent/boundue proprietary. NICHE / SMALL-CAP	Non-proprietary. MAIN STOCKS	Non-proprietary. NICHE / SMALL-CAP	Conferences	Equipment: proprietary	Equipment: non-proprietary	Access to IPOs	TOTAL

2.4 Can you provide extra detail on other services received that are currently bundled with execution?

2.1 Of the non-execution services you indicated at 2.h that: a) were over 2% of an averaged bundled trade and b) would incur a charge following the implementation of CP176 (either individually or as part of a bundle of non-execution services), please indicate:

- ..
- how the quality of these services to you may change: (taking into account any changes in quality) how the average price of these services to you may change (e.g. new price value of soft credits previously used): whether you will be able to pass the cost of these services through to your clients, and (taking into account be above) the quantity you would expect to purchase relative to the quantity currently obtained. . .

Where anticipated impacts of implementation are identical for different non-execution services, please aggregate these in a single table and state the services included in the table (e.g. if you believe the price, quantity and quality impact on two or more types of research listed above will be the same, these can be consolidated in the same table). Please provide any additional comments beneath each table.

	22
	3
	Ξ.
4	¥1
	č1
	ΞĮ
1	2

	Service 1	Service 2	Service 3	Serie 4	Service 5
Average quality	- up / down (same	wp / down (same	up.: down same	- up / down /ame	up / down hume
Averance melon	Philiphiano -	Philippine Street	("Cohores"	142 channel	PL channel
(cs10%)			and the second second		
Cast meavery	(% change)	(% change)	(*+ change)	1% change!	("is change)
(Aldwood 1000 24)					
Amount purchased	("= change)	("+ change)	("+ change)	("to change!	("he change!
(c.t. +10%)					

Comments:

	Service 1	Service 2	Service 3	Service 4	Survice 5
Average quality	up / down /same	up/down/same	up / down (same	up / down /same	up down sumo
				NAME OF TAXABLE AND ADDRESS OF TAXABLE ADDRESS OF T	
Average price	(%s change)	(*change)	(% change)	(+ change)	(%) (%) (%)
(# g -1054)		100 m			
Cost recovery.	("s change)	("s change!	("+ change)	("+ change)	(The change)
(eg. 100%+1600vaty)					
Amount purchased	("te change)	("se change)	(To change)	1% change!	("to change!
(50°+10%)					

Comments:

~	
×	
ā	
μ,	
L L	

	Service 1	Service 2	Service 3	Service 4	Service 5
Average quality	up / down/same	up / down /same	up / down /same	up / down /same	up / down 'same
Average price.	(%) change)	(% change)	1*+ change)	(1% change)	("is change)
(e.g10%)					
Cost recovery.	("In change)	(% change)	("+ change)	("* change)	(%» change [
(e.g. 100% (pcovery)			5 10 10 10 10 10 10 10 10 10 10 10 10 10		
Amount purchased	(% change)	("is change)	(atturne)	Pa changel	("schange!

Comments:

	i		
ł	ì		
		ļ	
	ļ	;	
	ļ		
i	i	ì	

Average quality ap / down /same up / down /same ap / down /same ap / down /same Average price (% change) (% change) (% change) (% change) (e.g10%) (% change) (% change) (% change) (% change) Cost recovery (% change) (% change) (% change) (% change) Avenue (% change) (% change) (% change) (% change)		Service 1	Service 2	Service 3	Service 4	Service 5
Average price 1% change) 1% c	Average quality	up/down/same	up / duwe same	up / thom isame	attirs/ tawop / dn	up / down same
(e.g. 4(9%)) Cost recovery (% change) (% change) (% change) (% change) (% change) c.g. 100% recovery) (% change) (% cha	Average price	("s change)	(% change)	("+ change)	("= change)	("= change)
Cost recovery (% change) (% chang	(t_0(- 2a)					Summer of the second
(e.g. 100% recovery) Amontonia (% change) (% change) (% change) (% change)	Cost recovery	(%) change)	1% change?	(** change)	1% change)	(% change)
Answurt purchased ("a change) ("a change) ("a change) ("a chu	(ABMORE 1001 27)					
0.4 a 1246 (1	Amount purchased	(% change)	(% change)	("Lichange)	("Inclumpe)	("is change)
eg. +10/sl	(+C+1001)					

Comments:

R. Research costs resulting from the implementation of CP176

R.a Can you categorise the types of research, as you understand them, by source (e.g. broker, independent researchers, etc.) and other appropriate classifications (e.g. proprietary / non-proprietary, verbal / written)?

R.b (In the light of the previous answer) Do you consider non-proprietary circulars sent by brokers and others to all clients to be research? Please explain your answer.

R.c - R.e Please populate the landscape tables relating to this question on pages 27 and 28

Instructions The tables ask about the production, consumption, price and value of research. The 1st table asks about research as currently produced, the 2nd table deals with research produced historically, and the third table asks you to predict production and consumption patterns if the proposals in CP176 are implemented. To guide the population of rows in the tables:

Volume produced/sourced (% of total): This is the quantity of research that you acquire from different sources, totalling to 100%

Volume used (% of total): This will differ from the above percentages if some of the research received is not used

Value (out of 100); Some types of research received will be valued more than others (e.g. in-house research may be thought to have a value of 90, but non-proprietary kroker research a value of 50). Level of production (relative to your needs): e.g. "noo little" or "too much" of each type of

Over or under-consumption in the market: In your opinion, would the market work better if some types of research were more heavily consumed a not consumed as much (e.g. answer "over" or 'under" where appropriate) research

Average price: Are some types of research over-priced affecting their consumption (e.g. state "too high" where appropriate) Please provide any additional comments relating to your population of the tables underneath the third table.

R.c. - R.e Please provide any additional comments relating to your population of the tables R.c. R.e.

က
\mathbf{X}
P
Ш
₽

R.f Please complete the table below to show:

- ..
- The value of research services received through soft credits: The percentage of services received that were purchased or provided by the issuer of the soft credits.

	2002	2001	2000
Value of soft credits spent on proprietury research (broker main stocks)			
the purchased or provided by broker that issued soft credits)			
Value of soft are dita spent on proprietary research (broker: niche : $SC)^{\alpha}$			
(%) purchased or provided by broker that issued soft credits)			
Value of soft credits spent on proprietary research (not, bouldque: man docks)			
(*+ purchased or provided by broker that issued soft credits)			
Value of soft credits spent on proprietary research ()mL houtique: mche / soft	3		
("a purchased or provided by broker that (smed soft credits)	0		
Value of soft areflets spent on non-proprietary research (main socies)			
C's purchased or provided by broker that issued soft condust			
Value of soft credits spem on non-proprietary research iniche (SC)			
(%) purchased or provided by broken that issued soft credits)			
Value of soft credits spent on other research			
(%) purchased or provided by broker that issued soft credits)			
Fotal value of soft credits spear on research			
Total % surchased or movided by budger that jound call credits			

* SC= Smill Cap

Please provide any comments relating to your population of the table.

R.i What percentage of research that you currently source is related to specialist or small-cap stocks?

R.j When assessing the impact of unbundling on the demand for research. Charles River Associates predict a "significant free rider problem" resulting from the "inability of fund managers to effectively discriminate between those clients that are willing to pay for research and those that are not." They conclude that this combined with "an unwillingness for any increase in annual management charges to relate the cost of advice" will mean that "the equilibrium level of research will fall below the optimum. "Please indicate your level of agreement with the above argument.

	Strongly Disagree	
	Divagree	
A DESCRIPTION OF A DESC	Assilier agree nor disagree	
	Agree	
	Strongly agree	

Please explain your answer:

R.k. Please indicate your level of agreement with the following statement from Oxera's Cost / Benefit analysis of CP176:

"Total spending on research may fall, but spending on in-house research, and the type of third party research currently paid for with hard money, is likely to increase proportionally, if not absolutely. As a result, general competitive conditions are likely to improve, which should be advantageous to the customers of research services."

Strongly Disagree	Disagree	Neither agree nor disagree	Agree	Strengly agree

Please explain your answer:

R.1.1.a How much do you currently spend on in-house research p.a. Please disaggregate your answer to include staff, space, materials and other costs.

R.1.2 If the proposals in CP176 are implemented, do you envisage setting up / increasing an inhouse research capability? (please state which and any measure of the scale of set-up or increase) R.1.4 If you do envisage setting up / enhancing your in-house research capability, can you list the incremental activities and costs involved in the set-up / enhangement of your in-house research ream?

R.1.5 Do you anticipate that these will result in the recruitment of new staff? If yes, please expand your mover to include numbers of staff / estimate of salaries. R.2.1 Netting your answers above in relation to research, how would you expect research on small cap stocks to be affected by a) further restrictions on softing and b) unbundling/rebating? Where possible, please provide evidence in support of your answer.

R.2.2 Please state your level of agreement with the following statement:

"#Row research is purchased with hard cash rather than hundled, the independent research server will be able to flowrish as it will compose on a level physing field and, in particular, he able to provide better quality research in small cap and niche stocks."

	ngly Disagree	
	Disagree Stro	
They goree har	disagree	
	Agree	
	Strongly agree	

Please explain your answer

R.2.4 Can you provide evidence from previous regulatory decisions / cost shocks in support of your views above?

R.3.2 If the services above that you currently receive with execution services were separated from these and purchased as a package of non-execution services what percentage of your expenditure on these services would you expect to recover from clients by way of increased manugement fee or explicit charges?

Ą

Please explain your answer

5

4. Process costs associated with unbundling, rebating and ending of softed ties	±.1 Cost of calculation of unbundled prices	4.1.1 Please populate the landscape table relating to this question on page 29.	Please complete the table listing any incremental activities and costs you anticipate will result from the need to calculate the price of currently bundled or sofied services with your current service providers.	4.1.2 Do you anticipate that these incremental activities and costs will result in the recruitment of new staff? If yes, please expand your answer to include numbers of staff' estimate of salaries		4.2 Cost of involcing	4.2.1 Can you describe any modifications that would need to be mude to your invoicing system to accommodate the proposals in CP176.			4.2.2 Please populate the landscape table relating to this question on page 29.	4.2.8 Do you inticipate that the above incremental activities and costs will result in the recruitment of new staff? If yes, please expand your answer to include numbers of staff / estimate of salaties.		4.2.9 To the extent that different contractual and invoicing procedures are currently operating for different trades (e.g. principal, agency, single stock, large / smull cap), how would calculation /	are pointed of piece and subsequent involving for uncontext services impose a significantly greater administrative / IT burden than negotiation of these and other trade types?	
3, Increased transaction costs arising from unbundling	3.1.1 If proposal 1 of CP176 is implemented, what incremental activities / procedures would you adoot to source equipment. research and conferences from new providence?	area provinsi and a superprovinsi and and and and and a supervised as a supervised of a supervised and a superv Mateory examples intervised and the supervised as a supervised as a supervised of a subsection of the supervised	were, negoriament of prives when present provinces as examinised in a taket second.	Hardware / software!	Valuatión & Performance Management services:		Other (if maternal):	3.1.2 If proposal 2 of CP176 is implemented, what incremental activities / procedures would you adopt to source equipment, research and conferences from new providers?	Note: negotiation of prices with present providers is examined in a later section.	Equipment:	Research	Conferences:	Other (if material):	3.1.4 Do you anticipate that the above incremental activities and costs will result in the recruitment of new staff? If yes, please expand your answer to include numbers of staff / estimate of salaries.	20

Γ

APPENDIX 3

က
×
P
Ē
đ
◄

4.2.10 Can you supply evidence relating to the last time that pricing mechanisms with brokers (and	4.3 Cost input of co.
it appressore, cuertus) cranageo signiticantly and the implications tuis had itt terms of the price calculation / negotiation and invoicing activities and costs as detailed above?	4.3.1 What is the VA1
Price calculation	
Invoicing	
The following questions should only be answered if, following the proposed implementation of CP176, you would continue to receive bundled execution and non-execution services and rebate the non-execution elements of these to clients' funds (see previous answer to question B.7).	4.3.2 Are these any w charge b) you would athice explicit posed
4.2.11 The cost of an unbundled trade is £8, of which the execution element can be valued at £5 and the non-execution element at £3. Your broker charges a bundled price of £7. How much would you observe to your effort to cover the nonzervention element?	

4.2.12 Do you currently pay out rebates to clients (e.g. to funds for miscalculated involces)? If so, approximately how many rebates do you issue a year? Can you quantify the costs of issuing these?

4.2.13 Please populate the landscape table relating to this question on page 30.

End of questions on rehating.

(3 Cost impact of extension of VAT to urbundled / unsofted services

4.3.1 What is the VAT impact from the urbuniling of non-execution services as you understand it?

4.3.3 Are these are ways in which at Fund Managers in general may be able to mitigate against this charge (b) yow would are/contenting against the charge (e.g. prior/continuous) structure / not arbite/explicit pass-through to chemis?"

4.3.3 Can you quantify any incremental activities / costs associated with the above methods of miligation? Can you explain how these are over and above activities normally undertaken?

4.3.4 Do you anticipate that the above interestal activities and costs will result in the recontinuent of new shaff? If yes, please expand your answer to include numbers of shaff? estimate of salaries

4.4 Cost impact of increased client management

4.4.1 Over and above client relationship activities and essets included above, do you anticipute any incremental client relationship activities and costs from the new regulation? Could you quantify these?

4.5 Cinst recovery 4.5.1 Thinking about costs associated with the incremental activities discussed above under the following headings:

- Transaction costs with new service providers
- Price re-negotiation with present providers
- Incremental invoicing costs Costs associated with the extension of VAT to non-execution services
 - Incremental chent management costs .

What percentage of these costs would you be able to pass through to clients by means of one or ·***** more explicit charges or raising your management fee?

Please explain your answer (include whether this percentage differs by type of client)

C. Other information

C.1 Please populate the landscape table relating to this question on page 31.

Instructions

Please complete the table as fully as possible. If you are unable to disaggregate to the types of clients included in the table, please supply aggregated information and indicate how this is aggregated. For fixed expenses, please consider only those costs whose level does not depend on the amount of business on the books. This will include overheads such as the cost of management, premises, basic audit fees, website development etc. The remaining expenses, such as salaries of non-managerial staff, mailings, telephone etc. whose consumption does depend on the amount of business, should be classified as variable

C.2 Please populate the landscape table relating to this question on page 32.

C.5 Please fill in the following table showing the unticipated effect on your revenues, gross margins, net margins, and market share if you were to raise management fees to clients by 5% , 10%, 3 basis points and 6 hasis points.

		Institution	ul clients			Retail	clients	
	Up 5%	Up 10%	Up 3bp	Up 6bp	Up 550	Lip 10%	Up 3bp	Up 6bp
Revenues								
Gross margin	400	0.00					A	
Net margin								
Market share								

Please explain your answer.

C.9 In the light of your analysis of the possible implications associated with implementation of the proposals contained in CP176, could you provide a description of the extra activities that you would undertake to mitigate against the possible extra costs associated with it?

C.10 Have you looked at the possibility of moving your Head Office or some all Fund Management divisions / activities away from the UK in the last 5 years? If so, what were the factors taken into account in this review and what were the conclusions?

C.11 In the light of your nawers above, can you provide any instances where you believe fewer activities will be undertaken / fewer costs incurred as a result of the implementation of CP176. Please quantify these if possible.

R.c Please complete the following table with reference to your current take-up of research services.

Current			Proprietar	~			1	Von-proprieto	(un		Total
vituation	In-house	Bruker main stocks	Broker niche / small cap	Ind./ boutique main stocks	Ind./ boutique niche/SC	Broker main stocks	Broker niche / small cap	Ind./ boutique main stocks	Ind./ houtique niche / SC	Others	
Volumes produced (sourced (% of total)											100%
Volumes used (% of total)											100%
Value (out of 100)											
Level of production (relative to your needs)							1-1				
Over- or under- consumption in the market											
Average price											

10 years ago			Proprietar					Ven-propriet	ary		Total
	In-house	Broker main stocks	Broker niche / small cap	Ind./ boutique main stocks	Ind./ boutique niche/SC	Broker main stocks	Broker niche / small cap	Ind./ beatique main stocko	Ind./ boutique miche / SC	Others	
Volumes produced / sourced (% of total)											100%
Volumes used % of total)											100%
Value cet of 100)											
relative to your needs)											
Over- or under- onsumption in the marke	t				1						
Average price											

If CP176 is	1000	20 M	Freprietar	y	APR STOR	Sec. 1	and a second	Non-propriet	ary.	11	Total
Implemented	in-house	Bruker main stocks	Broker niche / small cup	Ind./ boutique main stocks	Ind./ boutique niche / SC	Broker main stocks	Broker niche / small cap	Ind./ bourique main stocks	Ind./ boutique niche / SC	Others	
Volumes produced / sourced (% of lotal)											100%
Volumes used					:						94001
Value cet of 100)											- 19
Level of production relative to your needs)											
Over- or under- cossumption in the marke	14 14										
Average price											1.1

т т т

4.1.1 Please complete the table listing any incremental activities and costs you anticipate will result from the need to calculate the price of currently bundled or softed services with your current service providers.

Γ			
	Associated time and costs		
	Incremental activities		
	High-level activity		

4.2.2 Please complete the following table explaining any cost impact of incremental invoicing activities resulting from implementation of CP176

	Incremental administrative activities & costs	Incremental IT activities & costs
One-off modifications to present system		
Ongoing incremental administrative / system costs		
APPENDIX 3

4.2.13 Please complete the following table detailing incremental costs and activities arising from extra rebating to clients' funds.

	Incremental activities	Incremental costs
Additional administrative activities		
Additional IT / systems activities		
Other		

30

APPENDIX 3

TOTAL Other Overseas clients supermarkets Fund Direct Private chent UK retail (external) C.I Please complete the table below giving details of funds managed, fees and overheads chent (own) Private IFAs UK life JK pension Average initial commission levels respect of acquiring new business Variable expenses / overheads in respect of acquiring new business Variable expenses / overheads in Average funds lost / realised p/a 2 respect of ongoing business (E) respect of ongoing business (£) Fixed expenses / overheads in Fixed expenses / overheads in Funds under management (£) Average management fee for Annual new business (£) Average fee for new business Additional fees (%) e.g. existing business (%) performance related Number of clients (3) 20 9

(m)

APPENDIX 3

C.2 On average, what proportion of your clients have switched to / away from you in the last three years?

Other TOTAL	
Overseas clients	
	Fund
	Direct
UK retail	Private client
	Private client
	IFAs
UK life	
UK. pension	